Votes and Violence: Evidence from a Field Experiment in Nigeria^{*}

Paul Collier^{\dagger} and Pedro C. Vicente^{\ddagger}

First Draft: December, 2007; This Version: January, 2009

Abstract:

Following the wave of democratization during the 1990s, elections are now common in low-income societies. However, these elections are frequently flawed. We investigate the Nigerian general election of 2007, which is to date the largest election held in Africa and one seriously marred by violence. We designed and conducted a nationwide field experiment based on randomized anti-violence grassroots campaigning. We find direct effects on violence outcomes from exploring both subject-surveying and independent data sources. Crucially, we establish that voter intimidation is effective in reducing voter turnout, and that the violence was systematically dissociated from incumbents. We suggest that incumbents have a comparative advantage in alternative strategies, vote buying and ballot fraud. Voter intimidation may be a strategy of the weak analogous to terrorism.

JEL Codes: D72, D74, O55, P16.

Keywords: Violence, Conflict, Electoral Politics, Political Economy, Randomized Experiment, Field Experiment, Nigeria, West Africa

^{*} We wish to thank Christian Ahlin, Robert Bates, Catia Batista, Tim Besley, Michael Bratton, Michael Conlin, Marcel Fafchamps, Claudio Ferraz, Yvan Guichaoua, Masa Kudamatsu, Alice Mesnard, Abdul Mustapha, Luis Pinto, Laura Valderrama, and Eric Werker for helpful suggestions. We are particularly grateful to Ojobo Atukulu, Otive Igbuzor, and Olutayo Olujide at ActionAid International Nigeria, Austin Emeanua, campaigners Nwakaudu Chijoke Mark, Gbolahan Olubowale, George-Hill Anthony, Monday Itoghor, Umar Farouk, Emmanuel Nehemiah, Henry Mang and their field teams, and to the surveyors headed by Taofeeq Akinremi, Gbenga Adewunmi, Oluwasegun Olaniyan, and Moses Olusola: their professionalism, courage, and dedication to this project were truly outstanding. We also want to acknowledge the kind institutional collaboration of the Afrobarometer. We thank seminar participants at the LSE-Oxford iiG meetings, CSAE, NCDE, EEA, NEUDC, CEPR, AEA/ASSA conferences, and at numerous universities for useful comments. We wish to acknowledge financial support from the DfID - Department for International Development (UK), in the context of the iiG Consortium - 'Improving Institutions for Pro-Poor Growth'. All errors are our responsibility.

[†] University of Oxford and CEPR.

[‡] Trinity College Dublin, CSAE - University of Oxford, and BREAD.

"This election is a do-or-die affair." - President Olusegun Obasanjo, February 10th 2007

1 Introduction

The slow growth of Africa over the period since independence is now understood as being partly attributable to poor governance. Until the 1990s the predominant African political system was autocracy. As Besley and Kudamatsu (2007) show, while in some contexts autocracy has produced good economic performance, in Africa it has consistently been dysfunctional. During the 1990s many African autocracies were replaced by democracy, most dramatically in the region's largest society, Nigeria. Given the dismal record of autocracy, there was a reasonable expectation that democracy would achieve both accountability and legitimacy, and thereby both improve economic performance and reduce proneness to political violence. However, the record of elections in Africa and other recent low-income democracies is not encouraging. Kudamatsu (2006) measures government performance by infant mortality and shows that, in Africa, elections produce no improvement except in the rare instances in which the incumbent is defeated. Collier and Rohner (2008) find that, below per capita income of \$2,750, democracy significantly increases proneness to civil war and various other manifestations of violence, and Collier and Hoeffler (2008) find that in resource-rich economies such as Nigeria, electoral competition worsens economic performance unless combined with strong checks and balances.

We attempt to contribute to a better understanding of the surprising (non-)results of democracy in Africa by focusing on how elections have been conducted. Specifically, politicians may have been heavily reliant upon illegitimate strategies for getting elected. Indeed, numerous recent African elections aroused widespread international accusations that parties/candidates had resorted to miscounting of votes, bribery, and/or intimidation. Although these strategies may be inherently associated to the failure of democracy to improve government performance, our objective is to go well beyond the anecdotal: the comprehension of their specific causes and consequences may open new avenues for development policy intervention.

While we provide evidence on determinants of all three referred strategies, the main contribution of this paper is to use a field experiment to study the use of violence to intimidate voters. The context for our analysis is the 2007 Nigerian full round of national and state-level elections: this proved to be an all-too-suitable context for our purposes, as during the two days of these elections

over 300 people were killed. To the best of our knowledge this is the first paper formally analyzing electoral violence, and one of the few papers applying experimental methods to the study of elections in the developing world (the others being Wantchekon, 2003, and Vicente, 2007a). We investigate how intimidation changes voter behavior and which candidates are most identified with electoral violence.

To be able to unambiguously identify our causal relationships of interest we need exogenous variation in violence-related variables. To this purpose we designed and conducted a field experiment where a campaign against political violence was randomized across neighborhoods and villages of 6 states of Nigeria (representing all main socio-economic regions of the country). This campaign was conducted in half of those locations before the 2007 round of elections by a major international NGO, ActionAid, specializing on community participatory development. It included town meetings, popular theatres and the distribution of campaign material. It was aimed at empowering citizens to counteract local violence, and its activities were designed to reduce the costs of 'protest' and collective action. It also explicitly appealed to 'voting against violent politicians'.

Our measurement was based on a panel of 1149 survey respondents in all treatment and control areas of the experiment. These subjects were interviewed both before the anti-violence campaign and after the elections, and constituted the primary focus of campaigners. The interviews elicited a wide range of measurements of experience with and perceptions of violence (including specific incentive-compatible measurements). Subjects were also asked to report their intended and actual voting behavior (before and after the elections). Our survey-based violence variables were complemented by the compilation of actual violence measures from event-diaries for each location, kept by independent local journalists. Since there is a possibility of conformity bias from the treatment on subject-perceptions¹, we drew 300 additional respondents at the post-election stage, in treated areas only, who were not directly approached by the campaign. We also asked about the closest network of each respondent in the full sample. Both additional sets of data allow us to run robustness tests on the validity of subject-reports.

The main mechanism of action we propose in this experiment is therefore based on the effect of the campaign on grassroots' views and perceptions about violence (which ultimately caused less occurrences of violence at the local level). Crucially, the described primary exogenous effect of

¹ See Vicente (2007a) for a quantification of this potential bias.

the campaign enables us to answer our main empirical questions: on the effect of violence on voter turnout (by instrumenting violence with the randomized campaign), and on identifying the candidates that are most associated to violent behavior. Additionally, the effectiveness of the NGO campaign in reducing both the perception and the actual level of violence itself has notable policy implications.

Our most important empirical results point to violence as a systematic electoral strategy with systematic consequences for votes. Violent intimidation was identified to reduce voter turnout by a clear 10% for each 1/6 increase in electoral violence. And it was a strategy predominantly linked to non-incumbent groups (sometimes identified as the main political opposition, as is the case of AC in presidential elections). These results are based on robust effects of the campaign on violence outcomes, both using surveyed-subject reports and journalists' diaries. We also find that, in contrast with electoral violence, vote buying and ballot fraud tend to be associated with the incumbent and with more electorally-contested (swing) areas.

We interpret these results in the context of a specific model of electoral competition. There, both incumbent and challenger may intimidate voters towards abstention. In addition, only the incumbent may bribe voters to vote for him and may use ballot fraud to win the election. The incumbent, who always wins the election in equilibrium, only cares about winning; the Challenger only cares about his share of the popular vote. Swing voters contrast fear with money offered; base voters are assumed to condition their base support to a peaceful campaign by their candidate. In the model, intimidation comes from the Challenger when he is least competitive, while vote buying and fraud arise when the Challenger is most competitive.

In Section 2 we set out a theoretical framework of illegitimate election strategies. Section 3 describes the Nigerian context. Section 4 discusses the design of the experiment, a fuller description together with displays of campaign materials being given in Appendices 1 and 2. Section 5 gives an overview of the results, using descriptive statistics. Section 6 presents our core econometric results and offers robustness tests. Section 7 concludes with some implications for policy.

2. How to Win an Election: A Model of Electoral Violence, Vote Buying, and Fraud

Citizens are generally expected to use their vote or even lobby (Becker, 1983) to further their interests. However, there may be imperfections. Besley (2006) reviews the consequences if voters have poor information about government performance (analyzed in-depth by Grossman and Helpman, 1996, for a prominent example), or if the allegiance of many voters is predetermined by loyalties determined by identity. As information deteriorates and allegiance is frozen, a point is reached beyond which those potential politicians who are ill-motivated are not disciplined by the fear of losing votes and so enter politics. However, in elections analyzed by mainstream political economy the strategies open to candidates remain confined to those prevailing in the mature, high-income democracies: only strategies oriented to please regular citizens are considered. In many of the newly democratic low-income countries the only aspect of democracy that has been introduced is elections. There are neither 'checks-and-balances' upon the use of power, nor effective regulations for the conduct of the election. Indeed, in Nigeria, the introduction of elections in 1999 was preceded by 38 years of the postcolonial era of which 30 were spent under military ruling, and so no such regulations were in place.

In this context we propose an original model where electoral violence, vote buying and fraud are strategies available to politicians to help earning political power in elections. Note that among these strategies vote buying is the only one providing voters with a benefit. Let us describe our interpretation of these strategies in more detail.

The use of violence to intimidate voters may have clear advantages. If targeted on discouraging known opponents (e.g. as when allegiance is determined by identity) from voting, the pertinent voter behavior is highly observable (i.e. the polling station merely needs to be observed). Further, it is likely to be widely available to both incumbent and challenger: one hired gang can constitute a credible risk of violence to many voters. Vote buying suffers from the obvious limitation that if the ballot is secret it is difficult for the politician to enforce the bargain. It may nevertheless become effective either if the secrecy of the election is doubted, or if the voter attaches moral value to keeping her word. If we take vote buying as encompassing clientelism the enforcement problem is solved since 'payments' (e.g. public-sector jobs) will be conditional on being elected. These strategies are likely to advantage the incumbent, who is expected to have more money, may be suspected of being in a position to subvert the secrecy of the ballot, and is likely to be more convincing in proposing to 'clients' (sometimes by using resources from office holding). Finally, vote miscounting is expected to advantage the incumbent since the incumbent is more

likely to control the vote-counting process (e.g. the electoral commission). These are assumptions we take to our model.

We now present a formal electoral game where an incumbent I, a challenger C, and a continuum of voters of mass 1 interact to allocate political power. The timeline of the sequential complete information game is illustrated in Figure 1. It is related to Groseclose and Snyder (1996) who explain vote-buying behavior in the context of a sequential game. The Incumbent moves first by setting vote buying (or clientelism). Then, closer to the election, Incumbent and Challenger choose their levels of intimidation. Subsequently, voters decide whether to vote and for whom. Finally, the Incumbent may decide to use fraud to win elections (overriding the popular vote). Note that the asymmetry between Incumbent and Challenger on the range of strategies available (namely on vote buying and ballot fraud) is an assumption of the model, made for simplicity, that is tested in the empirical part of this paper.

Figure 1: The Sequence of the Game



Voters are defined on a continuum, with voter $i \in [0,1]$. For simplicity we assume that each candidate has a base vote of share b_j (j = I, C), with $0 < \lambda \le b_j \le 1$ (λ is the minimum base size, close to 0). We postulate that neither vote buying nor intimidation can affect opponent base voters' decisions on whether and for whom to vote. The remaining voters are 'swing' and drawn to the polls for private benefits or losses that are enforceable, as described next.

Voter *i* will vote for the Incumbent if

$$VB_i > I_i^C + I_i^I$$
,

with abstention chosen if

$$VB_i < I_{i}^C + I_i^I$$

For the case of swing-voter indifference,

i.e.
$$VB_i = I_{i}^C + I_i^I$$
,

which also constitutes our initial 'swing' status quo, we break the tie by punishing the candidate who has minimal base supporters. In other words, if $b_c = \lambda \wedge b_I > \lambda$, then indifference is broken in favor of the Incumbent; conversely, if $b_I = \lambda \wedge b_C > \lambda$, indifferent swing voters side with the Challenger. This assumption is meant to attribute a cost to extremist, 'anti-social' candidates, i.e. those holding the smallest base-voter share λ : when one such candidate is present, any swing voter at indifference is biased towards supporting the mainstream, alternative candidate. For symmetry, abstention is the decision taken by swing voters at indifference in case both candidates either have higher than minimal or minimal base supporters.

We need to make a further hypothesis regarding intimidation. Although intimidation may help the candidates by reducing the turnout of supporters of their respective contenders, we assume that it comes at a cost. As the intimidating candidate becomes identified with violence his committed, above-minimal base supporters are lost to the 'swing' mass, i.e. his support base is reduced, if larger, to λ which may be thought of as extremist support. Regarding base voters, we therefore assume an endemic distaste for violence.

We now turn to the payoffs of the candidates. We postulate there are two prizes to fight for: the executive prize, attributed to the candidate who wins the election (fixed), and a second-tier, political influence prize, which is allocated to the loser – this is measured by the share of the votes of the loser in the election (variable). This is because the loser is able to extract rents proportionately to his electoral performance². The candidates face the payoff

² This assumption could be interpreted in the context of a three-candidate game, where a third political player (a *bureaucracy*) would be inactive and entitled to obtain the second-tier residual power share, i.e. the Incumbent's share of voting. Still, we would be implicitly assuming the winning candidate does not care about second-tier political influence – a simplification in our game.

$$P_i - C_i$$
,

for j = I, C. Political benefits P are defined as

$$P_j \equiv aW_j + bS_j.$$

There, $W_j = 1$ in case *j* is the winner of the election - $W_j = 0$ otherwise. *j* may win the election either through fraud (but only if *j* is the Incumbent) or, in its absence, through a majority at the ballot, i.e. a share of the votes of $50\% + \varepsilon$ or higher. S_j is defined as the actual (ballot) voting share of candidate *j* if *j* is the loser of the election - $S_j = 0$ otherwise. We assume that winning the election is more attractive than all losing outcomes (a > b), and that both winning the election and earning votes as the loser confer large benefits relative to costs (a, b sufficiently high). Costs are defined as

$$C_I \equiv \int_0^1 (VB_i + I_i^I) di + F \text{ and } C_C \equiv \int_0^1 I_i^C di,$$

where VB is vote buying, I is intimidation, and F is the cost of fraud (F = 1 if the Incumbent uses fraud or F = 0 otherwise). We assume that the Incumbent has sufficient resources for ballot fraud always to be feasible, i.e. $C_I \le 1$ (where 1 is total resources available to the Incumbent), and that the Challenger is constrained by $C_C \le M_C$ (where $M_C > 0$ stands for total resources available to the Challenger).

2.1 Equilibrium

We now solve the game for the unique sub-game perfect equilibria, by analyzing the different parameter sets (i.e. values of b_1, b_c, M_c).

First, given the assumptions on the size of the budget of the Incumbent vs. the cost of ballot fraud, we know the Incumbent will, if necessary, always resort to fraud to win the election.

Hence, in equilibrium, given the assumptions on a and b the Incumbent always wins the election and the Challenger always maximizes his share of votes as the loser.

We begin by considering the case where $b_I > b_C$ and $b_C = \lambda$, i.e. where the Challenger is an extremist and the Incumbent has larger base support. As a dominant strategy, the Incumbent will want to spend nothing on vote buying, intimidation or fraud - he is already guaranteed to win the election $(b_I > b_C)$. Given this behavior, the Challenger will want swing voters to abstain rather than side with the Incumbent which they are inclined to do due to the Challenger's extremist position. By threatening intimidation the Challenger can achieve a discrete jump in his share of vote at negligible cost. We therefore observe intimidation from the Challenger (only) but neither vote buying nor fraud, the case being depicted in Figure 2.

Figure 2: Pure Intimidation



Now consider the situation in which $b_I > b_C$ and $b_C > \lambda$. In this parameter set it is still a dominant strategy for the Incumbent to be passive. Now however, were the Challenger to resort to intimidation, he would lose his base and so necessarily see his vote share decreased. This is the case of conventional politics: none of the illegitimate strategies - intimidation, vote buying and fraud – is used.

We now turn to cases where the Challenger's base is sufficiently large to pose a threat to the Incumbent's objective of winning the election $(b_I \le b_C)$.

We begin by considering $b_1 > \lambda$ and $b_c < 0.5$. Since $b_1 > \lambda$, swing voters abstain and there is no point in the Incumbent using intimidation (a strictly dominated strategy). Since a Challenger who resorted to intimidation would lose his above-minimal base and thereby receive a smaller share of votes, he fights a clean campaign. However, were the Incumbent to fight a clean campaign he would lose the election and so he has to choose among the remaining illegitimate strategies. We find the Incumbent resorts to vote-buying, purchasing just enough votes to take a majority ($b_c + \varepsilon$) – that will be cost-effective (relative to fraud) given negligible vote-buying spending. In this setting, we therefore see vote buying but neither intimidation nor fraud (see Figure 3). Note that $b_c < 0.5$ guarantees that the Incumbent may buy a majority.

Figure 3: Pure Vote-Buying I



If $b_I = \lambda$, the Incumbent is still obliged to choose at least one of the illegitimate strategies to win the election. We may have two outcomes depending upon the size of the Challenger's base (relative to majority) and the size of the Challenger's budget (relative to the Incumbent's), with the latter determining the relative attractiveness of vote buying and fraud for the Incumbent.

First suppose that $b_c < 0.5$ and $M_c < 1$. As before, $b_c < 0.5$ opens the possibility that the Incumbent buys a majority in equilibrium. Moreover, we know that for the Incumbent to win the election through the popular vote, he will have to induce some additional voters to vote for him: he cannot just rely upon intimidation. Suppose the Incumbent buys $b_c - \lambda + \varepsilon$ voters, paying just enough to get them to vote, with the remaining swing voters intimidated towards abstention or (indifferently for the Incumbent) bought. Then the best response by the Challenger would be to intimidate all non-base voters on the Incumbent's side towards abstention, yielding an electoral victory for the Challenger at negligible cost. The Incumbent must therefore lead the Challenger to consider spending his entire budget on intimidation – i.e. a race to the bottom. Since the Incumbent wants to win the election, he will buy (at least) the share of voters $0.5 - b_I + \varepsilon$ (ε in case $b_I = b_c = \lambda$) – otherwise the Challenger would remain passive (non-violent) and win the election. The Incumbent will race to the bottom by spending $M_c + \varepsilon$ with the referred voters (in any way - see Figure 4, VB^* , for an example), so that the Challenger will not be able to bring back all of them towards abstention. Crucially this amount is feasible under the condition $M_c < 1$, and in fact makes vote buying preferable to fraud (cost-minimizing). Note that, for the Incumbent, spending resources with intimidation (on any voter) is useless, as it does not oblige the Challenger to match these expenses when he responds. The Challenger will therefore be indifferent in terms of electoral results between intimidating and not intimidating. Hence, he will opt for the cheapest: clean campaigning. In this parameter set we therefore see only vote buying emerging in equilibrium - neither intimidation nor ballot fraud arise.





Now suppose instead that either $b_c \ge 0.5$ or both $b_c < 0.5$ and the maximum vote buying that the Incumbent can afford is weakly lower than what the Challenger can intimidate back (i.e. $M_c \ge 1$). If $b_c \ge 0.5$, clean campaigning is a dominant strategy for the Challenger. For $b_c < 0.5$ and $M_c \ge 1$, the Incumbent has insufficient resources to win a potential race to the bottom. Both possibilities mean the Incumbent cannot use vote buying to win the election, and so must resort to fraud. In this parameter set we therefore see only ballot fraud arising in equilibrium.

The model thus predicts that in situations where the Challenger is strong no violence will arise. In these cases the Incumbent will win the election by resorting to either vote buying or fraud. In contrast, when the Incumbent has sufficient support to win through an honest campaign, the Challenger may resort to violent intimidation, but he will do so only if he runs on marginal electoral support. In this model, the Incumbent never uses intimidation in equilibrium.

3. Background: The 2007 Nigerian Election

Nigeria is the most populous country in Africa, with estimated 135 million inhabitants in 2007³. Despite being a major oil producer, with the 10th largest oil proved reserves in the world (35b barrels)⁴, it ranks 201 in 233 countries in terms of GDP per capita (1400 USD PPP in 2005⁵). As implied by this failure to harness oil revenues for growth⁶, the quality of governance has been low: in Transparency International's Corruption Perception Index it ranks 147 in 179 countries (2007)⁷. In Chinua Achebe's (1983) words, 'the trouble with Nigeria is simply and squarely a failure of leadership'.

From 1999, with the passing of a new federal constitution, Nigeria moved to civilian rule⁸, under democratic elections: these happened in 1999, 2003, and 2007. However, all of these elections were damaged by widespread electoral malfeasance. By many accounts these elections were far from being 'free and fair'.

The election of 2007, which is the focus of our study, covered four distinct contests: presidential; federal house of representatives and senate; gubernatorial; and state assembly. Under Nigeria's federal constitution power is particularly concentrated in the president and the state governors.

³ CIA World Factbook.

⁴ Oil & Gas Journal, 103(47), December 19th, 2005.

⁵ World Development Indicators.

⁶ In this view, we believe the analysis of the Nigerian case also contributes to a better understanding of political incentives in resource-rich countries – and therefore of the 'natural resource curse' (Sachs and Warner, 1995; Mehlum et al , 2006; Robinson et al, 2006; Vicente, 2006, 2007b; Collier and Goderis, 2007).

 $^{^{7}}$ See Smith (2007), for a thorough account of the phenomenon of corruption in Nigeria.

⁸ See Maier (2000) for a description of this transfer of power and recent political history of Nigeria.

The 2007 election was not contested by the incumbent president, Obasanjo, due to his term limit. The key contestants were Umaru Yar'Adua, Muhammadu Buhari, and Atiku Abubakar. Yar'Adua was Obasanjo's chosen successor in the ruling PDP (People's Democracy Party). However, he was little known since until June 2006 Obasanjo had been hoping to change the constitution to permit a third term. Buhari had already been the main challenger in the 2003 election, and was standing for the ANPP (All Nigeria Peoples Party). A previous military ruler, his regime had been noted for a public campaign against corruption. Abubakar, although the incumbent Vice-President, was in serious conflict with President Obasanjo, and had been forced to switch party to the AC (Action Congress). Previously a customs officer with controversial sources of wealth, he had been indicted by the federal anti-corruption commission EFCC on multiple charges related with campaign fund embezzlement and bribery. At the core of the election campaign was the manifest determination of President Obasanjo to prevent Vice-President Abubakar from becoming the next president. In a phrase that became famous, Obasanjo described the election as a 'do-or-die affair'.

The ruling PDP duly won the election with 70% of votes, as did 28 of its candidates in the 36 gubernatorial elections. However, the election was deeply flawed through violence, bribery and vote miscounting. As illustration we present the assessments of three well-informed independent organizations:

'Rigging, violence and intimidation were so pervasive and on such naked display that they made a mockery of the electoral process. [...] Where voting did take place, many voters stayed away from the polls. [...] By the time voting ended, the body count had surpassed 300.' – Human Rights Watch.

'The irregularities were so numerous and so far-reaching that the election was a charade and did not meet the standards required for democratic elections.' Transition Monitoring Group (an NGO with 50,000 Nigerian observers).

'Nigeria's elections were not credible and fell far short of basic international standards. [...] Elections for president, state governors and legislators were marred by violence, poor organisation, lack of transparency, significant evidence of fraud, voter disenfranchisement and bias.' European Union Electoral Observation Mission.

These unfortunate features make the election well-suited for a study of electoral violence. In particular, violence may not have been simply a random spill-over from heightened antagonisms: it might have been used systematically as an electoral strategy.

4. The Design of the Experiment

To identify effects of violence on voting we need exogenous variation. Since the distribution of violence cannot be assumed to be random, our research design was centered around introducing a randomly distributed *reduction* in the perceived threat of violence and measure the consequences of this reduction. While to our knowledge this is the first application of experimental methods to the study of electoral violence, the method has already been demonstrated to be effective in other electoral contexts. Wantchekon (2003) pioneered the application to African elections, evaluating randomized political campaigns in Benin. He showed that for the incumbent a message of patronage to favored groups was more effective than one of national public goods. Vicente (2007a) comes closest to the current paper, analyzing an information campaign against vote buying practices⁹. Experimental methods have also been applied to more conventional election techniques (in American elections) such as canvassing, phone calls, and direct mail (Gerber and Green, 2000; Gerber, 2004).

Evidently, the most difficult part of our research design was to achieve a significant reduction in the perceived threat of violence in selected locations. As we will show, our intervention to reduce the perceived threat of violence was successful. However, necessarily our intervention could not *eliminate* the perceived threat: hence, our results provide merely a lower bound estimate of the full effects of voter intimidation on voter behavior.

We allied with the Nigerian chapter of a large and effective NGO, ActionAid, which regarded the prospect of political violence as a grave challenge to democracy and wished to counter it. ActionAid's input on designing a powerful campaign against political violence draw on its specialist expertise in community participatory development and its experienced field infrastructure. Its campaign included town meetings, street theatre, and the distribution of campaign materials all highly concentrated in particular locations. We provide details of the campaign in Appendix 1.

⁹ In related empirical work, though non-experimental, Brusco et al (2004) use survey methods to analyze vote buying in Argentina.

The campaign was designed to oppose voter intimidation through two distinct routes. The first was to neutralize intimidation by lowering the perceived threat to individual voters. The analytic foundation for this aspect of the campaign is the model of political protest of Kuran (1989, 1991). As exemplified by McMillan and Zoido (2004), a public call to a common protest action lowers its costs and so makes it easier to resist intimidation¹⁰. More specifically, if politician A threatens the known supporters of B, then the more B supporters who turn up to vote the lower the risk to each of them. At a minimum A's capacity for violence is spread over a larger group of potential victims, and quite possibly beyond some point the perpetrators of violence are themselves intimidated into inaction. The turnout of support for B will thus depend upon the expectation of each B supporter as to whether other supporters will ignore the threat. The opposition to intimidation consequently embeds a classic collective action problem. The key test of whether this first route was effective is thus whether it neutralized the turnout-reducing effect of intimidation among supporters of non-violent candidates. The second route by which the campaign aimed to oppose intimidation was to emphasize its lack of legitimacy. This was designed to make people who had intended to vote for those politicians who resorted to violence question whether their support was appropriate. In fact a key component of the slogan of the campaign was 'Vote against violent politicians'. We shall then see whether the campaign could persuade these supporters to switch their vote from violent candidates.

ActionAid agreed to implement the campaign in randomized locations. Given the exogeneity of this treatment, our impact measurement depended on two sources of information: first we conducted a panel survey of 1149 respondents in 24 enumeration areas, with rounds prior to and after the election; second we contracted independent journalists (one per enumeration area) to report/describe violent events that affected the neighborhood or village, through consultation of local bodies (e.g. town meetings, police) – 131 events were identified for the 24 locations, in the period before and after AAIN's campaign (2^{nd} semester 2006 to 2 weeks after the last April elections).

The baseline survey was conducted just before the campaign. The households in the survey were then targeted by the campaign in its wide range of door-to-door activities. The post-intervention

¹⁰ This idea also relates with theory of informational cascades by Bikhchandani et al (1992, 1998) and Lohmann (1994, 2000), which was proposed to explain the behavior of the masses.

survey was carried out after the elections, when results were known and post-election stability was achieved.

The survey was representative at the state level, being conducted in two states from each of the three large Nigerian regions (Southwest, Southeast, and North). See Appendix 1 for the details of our sampling framework. The campaign against violence was conducted in half of the enumeration areas, leaving the others as controls. Respondents were asked about violence, about their intended voting behavior, and about their actual voting behavior. Evidently, since our aim was to determine whether the reduction in perceived violence affected voter behavior, the design of the experiment depended upon two stages: that the campaign would reduce the perceived threat of violence, and that this would in turn affect voting. Hence, the surveys were designed to elicit evidence on each stage. The questions on violence were asked both prior to the campaign, focusing on a reference period ('the last year'), and after the campaign/elections, focusing on what had happened just before and during the elections (i.e. 'from January', when the baseline survey was in the field). The questions on voting were based on intentions (before) and self-reported actual decisions (after) regarding all the elections at stake in April 2007. See Appendix 1 for additional details of survey questionnaire design. In the figure below, we show the sequence of the experiment.

Figure 5: The Time Frame of the Experiment



Time Line

This design allows the use of a classic difference-in-differences econometric approach. In a first stage we investigate the effect of the anti-violence campaign on violence outcomes. In addition to

being of interest in its own right, this is a direct test of the mechanism by which the campaign might have changed voter behavior. The effect can be elicited through the specification:

$$VC_{ilt} = a + bX_{it} + cY_{l} + dt + eT_{l} + ft * T_{l} + \mathcal{E}_{ilt}, \qquad (1)$$

where VC is a violence or crime-related outcome, *i*, *l*, *t* are identifiers for individuals, locations, and time (before/after), *T* is a binary variable with value 1 for treated locations, *X* is a vector of controls (demographic, attitudinal), potentially time-varying, *Y* is a geographical fixed effect.

We can then run a reduced-form specification, yielding the effect of the treatment on voting behavior outcomes:

$$V_{ilt} = g + hX_{it} + jY_{l} + kt + mT_{l} + nt * T_{l} + \mathcal{E}_{ilt}, \qquad (2)$$

where V denotes a voting behavior measure (intended-before and actual reported-after).

We also display results for the estimation of

$$\Delta V_{il} = o + pX_i + qY_l + rVC_l + \mathcal{E}_{il}, \qquad (3)$$

where violence/crime is instrumented by the treatment variable (provided the identification of significant effects of the campaign in diminishing violence, i.e. from specification 1). This is the way by which we hope to be able to identify the effect of violence on voter turnout.

We use a rich set of individual control variables, including differences in information about candidates before and after the elections, campaign activism, and policy platforms perceived for the candidates and supported by respondents.

Finally, although we do not rely exclusively on survey respondents (also factual data from independent journalists) in our data design, survey measurement may generally have some limitations. In particular, since it is based on subject-reports, there is a possibility of encountering 'conformity biases': subjects may adapt their responses as to 'conform' to 'expected' effects of the treatment. To an extent we are defended from survey conformity biases because the survey

and the campaign were independent from each other, with distinct field teams and branding. Nevertheless, we attempt to verify the extent of this problem in three ways.

First we created an incentive-compatible individual measure of empowerment to counteract violence: we proposed an observable costly action against violence to all respondents in the survey (see the next section for details). Second, we compare panel respondents, all of whom had been directly approached by the campaign, with an oversample of 300 respondents in treated enumeration areas who were not directly approached by the campaign. We are thus able to contrast those respondents who were subject to the full range of interactions with the experiment (baseline interview, direct individualized campaign, post-intervention interview) with those who only face a post-intervention interview. Crucially, we ask common-knowledge questions, about their 'neighborhood or village'. If conformity bias is serious it should therefore be manifest in a difference between the responses of these two groups. Third, we identify the most closely connected respondents for each respondent in the panel survey. This information is used to correlate voting behavior within networks, and to employ these correlated measures of voting behavior instead of individual self-reports. This exercise allows us to control for measurement biases in voting behavior¹¹ that are present across networks.

5. Descriptive Statistics

We begin with an overview of how respondents perceived problems with the electoral process in each of the six states (Chart 1). We use a standardized subjective rating system, scaled 1-4 with higher scores indicating a more severe problem. All three of the illegitimate forms of voter influence were of concern to voters. In most states ballot fraud was perceived as the most serious problem, with the question being posed as 'How free and fair are the elections in terms of the 'counting of votes?'. Voter intimidation was usually the intermediate problem, the question being put forward specifically as a concern about 'security against violence originated by politicians'. Vote buying was rated as the least severe problem, although still prevalent, the question being

¹¹ On electoral behavior, we have to date been unable to gather fully disaggregated official electoral results and it seems unlikely that they exist for all locations. Results were announced in terms of the overall totals in a process that appears to have by-passed the need to aggregate actual votes. Note nevertheless that Vicente (2007a) was able to contrast self reports to disaggregated electoral results in Sao Tome and Principe's 2006 presidential election, without significant differences encountered. Although different, Nigeria and Sao Tome and Principe are neighbouring countries, which gives us some assurance that self reports may be an adequate source of voting data in that broader geographical region.





Evidently our approach depends upon the effectiveness of the anti-violence campaign. While we investigate this more formally in the next section, here we provide an overview. Respondents in the panel were asked the same questions twice, once referring to the period prior to the campaign and then referring to the period after it. Respondents in both the treatment and control areas might change their answers due to experiences during the intervening period, but only in the treated areas would respondents have encountered the campaign. We therefore focus on whether the change in responses differs as between treatment and control areas. First, consider four measures of perceived threats: general violence within the community, gang violence, physical intimidation (as a pure crime measure), and violence intensity (classified from journalists diary-descriptions into a 1-5 scale). For all four measures the treatment areas have a larger reduction in perceived threat than the control areas. Further, the two types of perceived violence most pertinent for voter intimidation, gangs and physical intimidation, show large falls in the treatment areas but are virtually unchanged in the control areas. Next, consider the obverse of perceived threats: three measures of confidence in electoral security. Respondents were asked about security, empowerment, and knowledge of how to resist violence. All three measures show larger improvements for the treatment areas than the controls, and again the most pertinent measure, security, shows a large improvement for the treatment areas but virtually no change for the controls.

Since talk is cheap, as referred, we supplemented the survey with a direct measure of an incentive-compatible action. All respondents were given a postcard which they could choose whether to post. On the card there was a message demanding that more attention be paid to countering voter intimidation. Since in order to post the card the respondent had to go to a post-office, this was a costly action. The respondent was more likely to incur this cost the stronger was her sense that intimidation should/could be countered – see Appendix 1 for the full description of this variable's design. Thus, differences in the response rate between treated and control areas are a useful measure of whether the campaign was effective. As shown in Chart 2, in both areas response rates were remarkably high: around a third of respondents returned the card, this being consistent with a high degree of concern about the problem. But response rates were even higher in treated areas than in their controls.



Since there is some basis for regarding the campaign as effective, the consequential question arises as to whether the perceived reduction in the threat of violence in the treated areas actually changed voting behavior. Again, the two-round structure of the survey is helpful in that it enables us to focus directly on the change between stated intentions during the first round and actual voting behavior as reported during the second round (see Chart 3).

First, consider voter turnout. Unsurprisingly, in both treated and control areas the election campaign succeeded in mobilizing people to vote: many people who in the first round of the

survey had stated that they did not intend to vote ended up voting. In the treatment areas this effect was clearly larger. The difference is consistent with the hypothesis that citizens who had been intimidated into deciding not to vote found the 'courage' to do so as a result of a perceived reduction in threat generated by the campaign. However, we are particularly concerned to see whether voter intimidation was used strategically by politicians. Hence, we are interested in the effect of the anti-violence campaign on the supporters of each party/candidate individually. There is a clear pattern. The incumbent parties, both at the presidential and gubernatorial levels, mainly PDP, clearly gained as a result of the campaign. Supporters who had intended not to vote appear to have been empowered by the campaign to turn out and vote for incumbents. However, the campaign had an objective beyond its core intent to empower people to resist being intimidated into not voting. The appeal to 'vote against violent politicians' aimed to weaken support for candidates who supported violence. Respondents initially intending to vote for a politician perceived as violent may have decided either to abstain or even to switch to other candidates. Most clearly at the presidential level, AC lost support. This is consistent with instability being a strategy of political desperation: Abubakar, the leader of the AC, had neither the advantages of the incumbent PDP, nor of the ANPP as the northern established opposition party.



Indeed, for both the presidential and gubernatorial elections, we find evidence consistent with both objectives of the campaign. PDP supporters were more likely to switch from an intention not to vote to actually voting in treatment areas. In contrast, AC supporters who had initially intended to vote were more likely to abstain in treatment areas, and this abstention effect was reinforced by

vote switching: AC supporters were also more likely to switch to the PDP in treatment locations. We revisit this decomposition in the next section.

6. Econometric Results

We begin by exploring whether the three illegitimate electoral methods, violence, vote buying, and ballot fraud, are allocated systematically. These results do not use the anti-violence experimental design; they target location-wide determinants of the perceived importance of these methods in the late stages of the 2007 election campaign. The results are presented in Table A1 in Appendix 3. We begin with a simple specification in which the explanatory variables are the expected tightness of the election, and whether the ruling PDP is incumbent in the state. Tightness is proxied by the absolute difference between incumbent and opposition scores at the location level in the previous gubernatorial¹² elections in 2003. We also control for whether the enumeration area is urban although in the event this is only significant for violent intimidation. The OLS results find that the tighter is the election the greater are the perceived problems of vote buying and ballot fraud, the two strategies that we have suggested are best-suited to the incumbent. In contrast, violence is significantly reduced if the race is competitive. The effects are large: treating the four-point scale of responses as linear, in tight races vote buying is perceived as being 41% greater¹³ and electoral fraud 53% more frequent. In contrast, violence is perceived as being far less of a problem in tight races, with a reduction of 25%. All these effects are statistically significant at the 1% level while clustering standard errors at the enumeration area level. This is evidence consistent with the idea that political expenditures on vote buying and fraud are used strategically when the election is competitive. In contrast, violence may be an instrument used for making a political stand by weaker parties or groups when they do not hold significant base support (analogously to terrorism). We continue exploring this hypothesis in the remaining of the paper.

Controlling for the tightness of the race, PDP incumbency at the state level increases the perceived problem of ballot fraud (by 15%), consistent with the hypothesis that for this strategy a combination of national and local incumbency is needed.

¹² The use of gubernatorial election scores reinforces the local nature of political competition we want to capture.

¹³ The same result appears in Vicente (2007a), consistently with classical swing voter theory (e.g. Dixit and Londregan, 1996), regarding the use of vote buying in Sao Tome and Principe.

We next focus more closely upon the chosen strategies by the PDP. While PDP as the national incumbent evidently had the advantage in terms of ballot fraud, for the other two strategies respondents were asked which party they regarded as the main perpetrator. To investigate the strategic use by the PDP of these other methods we create a variable which measures for each method the prominence of the PDP relative to other parties in these reports. We then weight the incidence of each method by this ratio. With this new dependent variable we find that in locations where the PDP is the incumbent it is less likely to resort to violence but more likely to resort to vote buying. All effects are statistically significant at least at the 5% level whether we control only for state dummy variables, or include basic infrastructural characteristics of the sites¹⁴ and individual demographic and attitudinal controls¹⁵.

We also present a very similar exercise using the nationally-representative, 37-state sample of the 2007 pre-election Afrobarometer survey. While the main dependent variables are the same, the subjacent questions were asked in the pre-election period. We find a very similar pattern, which allows us to state that our experimental sites do not differ much in terms of electoral strategies from a nationally-representative sample.

The core of our analysis is on the effects of electoral violence on voting and on identifying 'violent politicians', and for these we need to rely upon the anti-violence campaign. This requires two preliminary steps: the demonstration that the treatment is indeed randomized, and the demonstration that the campaign was effective in changing the perceived threat of political violence.

In Appendix, Tables A2, we compare the characteristics of the treated and control groups: their demographic profiles, baseline violence outcomes, and baseline self-reported electoral preferences. Since these variables are unaffected by the intervention, any differences between treatment and control are a product of luck. We generally find no statistically-relevant differences (at standard levels) between treatment and control groups. The exception is that the control group

¹⁴ These were chosen from a list of variables concerning the enumeration areas: existence of post office, school, police station, electricity grid, piped water, sewage system, health clinic, recreational facilities, places of worship, town halls or community buildings, market stalls.

¹⁵ Demographic controls are chosen from a wide range of variables: gender, age, household head and size, marital status, ethnic group and language, religion (faith, intensity), schooling, job status, occupation, property, household expenditure and welfare, health status. Attitudinal controls include: consistency measures, interest in public affairs, media exposure, campaign activism, knowledge about candidacies, policy (perceptions about candidates, own preferences).

seems to have better knowledge of ways to counteract violence (+9%). This is therefore evidence that the randomization was generally effective in gathering comparable groups of locations and respondents.

Table A3 in Appendix reports our regressions testing whether the campaign was effective on changing violence-related measures. We use a wide range of perception and experience variables from our survey measurement. The first set concerns general political freedoms: on 'thinking', on 'joining a party', on 'voting freely', on 'being free from insecurity', and on the perceived fairness of elections, in general and as related to violence. We then focus on general perceptions at the local level: conflict ('within the local community' contrasted to 'within family'), intimidation ('threatening negative consequences in order to induce voting in a certain way'), and electoral violence ('general violence related to politics'). We distinguish between violence from the top and empowerment against/sympathy for violence at the bottom. For the first set of variables, we display perceptions concerning 'influence of political assassinations on instilling a climate of fear' (frequent in the 2006 primaries), politicians 'openly advocating violence', 'violent gangs being active', and 'security from violence originated by politicians'. For the second set, we analyze 'support for do-or-die affair', 'local population's standing against violence originated by politicians', 'empowerment' and 'knowledge of ways to resist violence', as well as the postcard variable, and sympathy for violence. The final set of survey violence measurements comes from a batch of standard questions (both perceptions and experience) on local crime (which could be indirectly related to politics – namely through gang activities): 'thefts in public places', 'purposely-made damages to property (vandalism)', 'physical threats/intimidation', and policeinduced problems.

We use specifications with classical difference-in-differences (regressions on the levels – specification 1 above). Where level-measurement is not available, namely for questions asked only in the post-election survey, we rely upon time differences (gathered from the post-election survey) regressed on treatment. The first specification for each outcome variable is run with state dummy variables to control for geographical fixed effects, while the second adds controls for individual demographic and attitudinal characteristics. We use OLS regressions for all outcomes¹⁶

¹⁶ Although these outcomes are based on subjective scales, provided the way these scales were handled during the fieldwork (described in the Appendix 1), we are secure that assuming linear distances between the different levels is appropriate. As a robustness test, we did run Ordered Probit regressions - these generally showed no relevant difference to OLS estimates.

except for the postcard variable, where we use Probit. We cluster standard errors at the level of the enumeration area.

Overall we found clear effects of the campaign in most violence-related measures, the results being robust both in terms of magnitude and statistical significance to the inclusion of individual controls. The campaign changed responses on political freedoms by 9-12%, and those on free and fair elections by 18-21%. In terms of general electoral violence at the local level we found: for 'conflict within the local community' there was a 12% effect; in contrast, there was no significant effect on perceived conflict *within* families, which was, of course, not an objective of the campaign; we observe a 5% change in intimidation directly conditional on voting.

On local violence originated at the top, effects ranged from a 6% change in the influence of assassinations, an 8% change in politicians advocating violence and in gang activity, to a 10% change in security from violence originated by politicians.

On empowerment against/support for violence at the local level, we find very clear effects: 12% on support for 'do-or-die' at the grassroots, 8% on knowledge of ways to counteract violence. Recall that in addition to these perceptions the research design included an incentive-compatible action: whether the respondent went to the post office to send an anti-violence postcard. We thus also analyze effects of the campaign on the postcard variable, a measure that we interpret as empowerment to counteract violence. Treated respondents were found to send the postcard 14% more frequently than their control counterparts, with statistical significance at the 10% level.

We now turn to our survey measures of crime in the locality. The campaign was designed primarily to increase people's willingness to withstand intimidation, rather than actually to reduce related forms of criminality. We generally find no significant effects on experience with crime but significant effects on perceptions of crime in its different forms¹⁷.

In Table A4 (Appendix), we show effects of the treatment on intensity of electoral violence, as reported by independent local journalists (based on information gathered from direct observation

¹⁷ We did however find a clear 5% decrease on perceptions of police-induced problems as a result of the campaign, i.e. in the same direction as most other violence outcomes. This finding reassures us that the campaign does not seem to have been understood by respondents as biased in favor of the incumbent (who controls the police), as one would expect from the independent nature of the campaign sponsor (international NGO ActionAid).

and local institutions like police and town meetings). We find a 13% reduction in the intensity of electoral violence, significant at the 10% level. Once we add incidence to our main regression, we keep the same sign but lose the statistical significance. The clear effect on incidence constitutes evidence that there was an effect on perpetration of (actual) violence. Indeed, the induced reduction in the effectiveness of intimidation (the direct aim of the campaign) could lead to a reduction in the actual level of electoral violence as politicians adjusted their strategies. In the event, although we find that the campaign generally did not have significant effects on actual (experienced) general crime, we find some evidence that actual electoral violence diminished in intensity. This fact helps us to interpret our survey-based results away from pure conformity-like effects.

We now turn to the effects of our randomized campaign on the electoral behavior of our panel of respondents. Tables A5 in Appendix show the regressions of electoral behavior on the treatment (key regressions are difference-in-differences on the levels – specification 2). Since all our dependent variables are binary we use Probit estimations. As before, for each outcome variable, we begin by showing the simple difference-in-differences regression with geographical fixed effects; then we add controls for individual demographic and attitudinal characteristics. All regressions allow for correlation of errors at the enumeration area level. We focus on voter turnout and voting patterns in the presidential and gubernatorial elections, these being the elections where the stakes were highest (where the executive powers are concentrated). As in Chart 3 we compare intended behavior with reported actual behavior.

One of our key results concerns the effect of violence on turnout. Recall that we measure the change from stated intentions to vote to actual behavior as reported after the election and that overall almost a quarter of those who voted had done so despite a prior intention not to vote. The main purpose of the campaign was to persuade people who had decided not to vote because of intimidation to vote after all. Since we can already reasonably conclude that the campaign achieved its intended effect, we can interpret this effect of the campaign on turnout as being qualitatively the opposite of the effect of electoral violence itself¹⁸.

¹⁸ However, the campaign also had a secondary purpose: to turn violence into a vote-loser for those politicians who espoused it. While the former effect would raise the turnout of those who had been intimidated, the latter effect could reduce the turnout of those who had planned to support candidates who espoused violence. Hence, the change in overall turnout is the net result of these two offsetting effects.

We find that the primary purpose of the campaign succeeded: the proportion of people who changed their mind and voted was 9% larger in the treated group, an effect that is significant at the 10% level. Since the campaign merely reduced rather than eliminated the threat of violence it is a lower bound estimate of the effect of violence. Indeed, we can go beyond this lower-bound estimate as follows. We proxy violence by our most specific measure, namely 'security from violence originated by politicians at the local level', defined on a 6-point scale, and estimate its effect on voter turnout, instrumenting 'security' with the treatment as in specification 3 above. With this IV approach we target the effect of violence on turnout, a much larger effect: turnout increases by 10% for each unit of security. This suggests that electoral violence was an effective strategy in keeping those likely to vote for opponents away from the polls. In turn, this may help to account for why it was used by politicians.

We now turn to the effects of the campaign on each party's score. We find that in the presidential election the campaign reduced the vote for the AC by 4%, an effect significant at the 5% level. Recall that the AC was the weakest of the three parties (nationally) and its presidential candidate was portrayed in the media as espousing instability. The reduced vote for the AC is evidence that the secondary objective of the campaign also seems to have worked: people who had initially planned to support the AC decided to punish it whether by abstention or switching their vote to other parties. The other two main presidential candidates' scores are not significantly changed by the campaign.

At the gubernatorial elections the campaign increased the vote of the PDP by 13%, an effect significant at the 5% level. However, we might expect violence in the gubernatorial elections to vary state-by-state. Even if violence is a strategy of the weak, not all weak gubernatorial candidates might resort to it. We certainly find substantial variation: the overall effects are due to only three of the six states. In Rivers the PDP gained as a result of the campaign by a clear 25%, significant at 1%. In Oyo and Plateau the PDP gained almost as much, by 21% and 19% respectively. In all three states the PDP was the incumbent. We do not find significant effects for second and third parties at the state level¹⁹.

¹⁹ Indeed, several sources point to the importance of electoral violence by marginal groups without a clear affiliation to the main parties. For instance, in Oyo state, Human Rights Watch underlined the role of violent groups within PDP, defeated in its primary elections. For further details, see 'Criminal Politics: Violence, 'Godfathers', and Corruption in Nigeria', October 2007. In addition, the International Foundation for Electoral Systems (IFES), who implemented nationwide surveys during the 2007 Nigerian elections, considers 40% of the electoral violence to be originating outside the main parties, PDP, AC, and ANPP ('A Nigerian Perspective on the 2007 Presidential and Parliamentary Elections', August 2007).

In the final part of Table 4, we test the effects of the treatment on the specific patterns of vote change for the two parties that seem to be identified above as the ones that concentrated experimental action (PDP and AC). Namely, we test whether the treatment led voters to move from AC to abstention and to PDP (presidential elections), and to move to PDP from abstention and from AC (gubernatorial elections). We find evidence consistent with all these effects, although only moves from and to abstention are statistically significant.

6.1 Robustness

We now undertake three sets of robustness tests: the first regards testing for 'conformity biases' in survey data, both for violence outcomes and for voting behavior measures; the second asks whether there is a specific demographic profile that was particularly affected by the campaign both in respect of its attitudes towards violence and its voting behavior; the third tests for contamination of control areas by the treatment.

On conformity, we may still be concerned that our survey measurement of violence and voting behavior outcomes may be affected by a tendency of respondents to adapt to specific survey conditions (potentially perceived as) correlated with the treatment.

First we test the robustness of the effects of the treatment on violence outcomes – the direct mechanism of our experiment. We compare, within treatment areas (only), panel respondents, who faced the whole experimental machinery, namely two rounds of the survey and direct contact with campaigners, with the post-election oversample subjects. The subjects of the oversample were approached only for the second round of the survey and had no direct contact with campaigners, although they may still have seen the street activities and been generally aware (through their network) of the campaign. We assume that oversample respondents are less constrained to conform than panel subjects.

Table A6 in Appendix 3 reports results of difference-in-differences regressions for the questions in Table A3 asked both to the core respondents and to the oversample. For most measures there are no statistically significant effects of being approached directly. Of course, even were there to be differences these might be due to the effectiveness of direct contact rather than to higher conformity by the panel respondents. This is clearly the case for knowledge of ways to counteract

violence (significant at the 10% level). However, since overall the differences are negligible we can reasonably reaffirm that conformity biases were not severe.

Second we direct efforts into assessing the solidity of our effects on voting behavior. Here we use a thorough module of our survey measurement on social networks. We asked each respondent about 'regularly chatting with' and about 'family links to' each of the other subject households in our survey (including the treatment oversample). Moreover, we collected, through detailed enumeration area maps, the physical distances between each subject household at each survey location. These three measures of connectedness between the elements of our sample are used to identify, for each subject, the other *n* closest subjects. This information is used to correlate the voting behavior of the respondents with that of their close networks. Fitted values instead of actual responses per subject are used in the dependent variable to assess the robustness of the effects of the treatment. This exercise clearly assumes that true voting behavior is correlated within a network but that measurement biases are not. If the treatment effects are maintained, which we show is the case, we may be more secure in dismissing the possibility of measurement biases that are correlated across networks in the original data (which we hope to take out by performing this exercise).

We find in Table A7 (Appendix) that indeed all main treatment effects, on increasing voter turnout, decreasing AC's performance in the presidential election, and on benefitting the incumbents in the gubernatorial election, are maintained. We focus on 'chatting' as our preferred measure of connectedness of respondents, but also show results using all three alternative variables. The physical distance measure (identifying closest links for almost all respondents) allows us to increase the number of observations in the regressions. However it also adds some noise (since our geographical measure embeds many 'close' relationships that do not mean real interaction), which generally translates into larger standard errors in the estimates.

We next check whether there is a particular demographic profile that was particularly affected by the campaign: both attitudes and voting behavior changing at the same time (Table A8 in Appendix). For this we focus on those dependent variables where we have already found a clear effect and introduce demographic variables as interactions with the explanatory variables deployed previously. Note that for changes in voting the dependent variable is defined on the range 1, 0, or -1, corresponding to a change towards party X, no change, and a change from party X, respectively: we therefore use Ordered Probit. We found that the characteristics most

associated with both a change in perceptions of violence and in voting behavior in response to the campaign were respondents in smaller households, working outside the home, not owning land, and of Yoruba origin. The fact that our results rest upon an identifiable group of people who were particularly susceptible to the campaign strengthens confidence in our interpretation of the mechanism: a change in attitudes leading to a change in behavior.

Finally we test for contamination of control areas (Table A9 in Appendix) by regressing our main outcomes of interest (on violence and on voting behavior) on distance to closest treatment area, while using control location observations only. Although for most regressions we do not find significant effects, there seems to be some evidence of contamination for intensity of violence (journals measurement), empowerment, and knowledge of ways to counteract violence. This leads us to conclude that the treatment effects estimated earlier regarding these outcomes may be underestimated.

7. Conclusion

Since the fall of the Soviet Union there has been a proliferation of elections in societies with weak governance. In many of these societies electoral competition has produced serious levels of violence. Our study is the first to have investigated the causes and consequences of electoral violence using experimental methods. The context for our experiment, the Nigerian presidential elections of 2007, was the largest African election to date; it was fought over the highest stakes in Africa: political control of Nigerian oil revenues; and it was indeed violent. We have shown that this violence systematically reduced voter turnout. More surprisingly, it was systematically associated with non-incumbent groups. The two other important illegitimate strategies of gaining votes, ballot fraud and vote-buying, were both rife alongside violence, but these were employed quite differently. They seem to have been used predominantly by the incumbent party and deployed most vigorously where the electoral contest was expected to be particularly tight. This pattern is consistent with the incumbent party having an absolute and comparative advantage in ballot fraud, due to control of the count, and in vote buying, due to the scope for embezzlement of public resources. All three illegitimate methods were used strategically. Ballot fraud and vote buying were election-winning strategies whereas political violence appears to have been a strategy of desperation by the weak and thus somewhat analogous to terrorism.

Finally, the success of the campaign suggests that violence can be countered. This is of direct importance: in both the Nigerian and the (also prominent) Kenyan elections of 2007 several hundred lives were lost. If violence is used by candidates who cannot win, what is its rationale? Our best hypothesis is that it strengthens the candidate's post-election position. However, electoral violence is unlike terrorism in one important respect. With terrorism the violence is the signal of power, whereas with electoral violence the *election result* is the signal: violence is used instrumentally to improve the result. Voter intimidation works by lowering the turnout for other candidates, thereby increasing the share of the violent candidate. We have demonstrated that the primary component of our campaign succeeded in countering the reduction in turnout. It did this by emboldening people not to be intimidated by threats. However, a campaign aimed only at emboldening people cannot hope to *eliminate* the power of threats to intimidate. Thus, this objective of the campaign can merely make voter intimidation less effective. The induced response of those politicians using violence may be to divert effort to other strategies, but it could instead be to increase threats so as to counter the reduced efficiency. This is the significance of the other component of the campaign: to reduce the vote of violent candidates. This also succeeded. Unlike the anti-intimidation message, the message not to vote for violent candidates has the potential to make violence counter-productive. We have demonstrated that a dual-purpose campaign can successfully deliver both messages. An implication is that political violence can be effectively countered by a rather straightforward measure.

References

Achebe, Chinua (1983), The Trouble with Nigeria, Heinemann Educational Publishers;

- Bikhchandani, Sushil, David Hirshleifer, and Ivo Welch (1992), A Theory of Fads, Fashion, Custom, and Cultural Change as Informational Cascades, Journal of Political Economy, 100(5), pp. 992-1026;
- Bikhchandani, Sushil, David Hirshleifer, and Ivo Welch (1998), *Learning from the Behavior of Others: Conformity, Fads, and Informational Cascades*, Journal of Economic Perspectives, 12(3), pp. 151-170;
- Becker, Gary S. (1983), A Theory of Competition among Pressure Groups for Political Influence, Quarterly Journal of Economics, 98(3), pp. 371-400;
- Besley, Timothy (2006), *Principled Agents? The Political Economy of Good Government*, Oxford University Press;
- Besley, Timothy, and Masayuki Kudamatsu (2007), *Making Autocracy Work*, CEPR Discussion Paper 6371;
- Brusco, Valeria, Marcelo Nazareno, and Susan C. Stokes (2004), *Vote Buying in Argentina*, Latin American Research Review, 39(2), pp. 66-88;
- Collier, Paul and Benedikt Goderis (2007), Commodity Prices, Growth, and the Natural Resource Curse: Reconciling a Conundrum, Oxford University, Working Paper.
- Collier, Paul and Anke Hoeffler (2008), *Testing the Neo-Con Agenda: Democracy in Resource-Rich Societies*, European Economic Review, forthcoming;
- Collier, Paul and Dominic Rohner (2008), Democracy, Development, and Conflict, *Journal of the European Economic Association*, 6(2-3): 531-540.
- Dixit, Avinash, and John Londregan (1996), *The Determinants of Special Interests in Redistributive Politics*, Journal of Politics, 58(4), pp. 1132-1155;
- Gerber, Alan S. (2004), *Does Campaign Spending Work? Field Experiments Provide Evidence* and Suggest New Theory, American Behavioral Scientist, 47(5), pp. 541-574;
- Gerber, Alan S. and Donald P. Green (2000), The Effects of Canvassing, Telephone Calls, and Direct Mail on Voter Turnout: A Field Experiment, American Political Science Review, 94(3), pp. 653-663;
- Groseclose, Tim, and James M. Snyder, Jr. (1996), *Buying Supermajorities*, American Political Science Review, 90(2), pp. 303-315;
- Grossman, Gene M., and Elhanan Helpman (1996), *Electoral Competition and Special Interest Politics*, Review of Economic Studies, 63(2), pp. 265-286;
- Kudamatsu, Masayuki (2006), *Has Democratization Reduced Infant Mortality in Sub-Saharan Africa? Evidence from Micro Data*, University of Stockholm, IIES, Working Paper;
- Kuran, Timur (1989), Sparks and Prairie Fires: A Theory of Unanticipated Political Revolution, Public Choice, 61, pp. 41-74;
- Kuran, Timur (1991), *The East European Revolution of 1989: Is it Surprising that We Were Surprised?*, American Economic Review Papers and Proceedings, 81(2), pp. 121-125;

- Lohmann, Susanne (1994), The Dynamics of Informational Cascades: The Monday Demonstrations in Leipzig, East Germany, 1989-91, World Politics, 47(1), pp. 42-101;
- Lohmann, Susanne (2000), *Collective Action Cascades: An Informational Rational for the Power in Numbers*, Journal of Economic Surveys, 14(5), pp. 655-684;
- Maier, Karl (2000), This House Has Fallen: Nigeria in Crisis, Westview Press;
- Mehlum, Halvor, Karl Moene and Ragnar Torvik (2006), *Institutions and the Resource Curse*, The Economic Journal, 116(508), pp. 1-20;
- McMillan, John and Pablo Zoido (2004), *How to Subvert Democracy: Montesinos in Peru*, Journal of Economic Perspectives, 18(4), pp 69–92;
- Robinson, James A., Ragnar Torvik and Thierry Verdier (2006), *Political Foundations of the Resource Curse*, Journal of Development Economics, 79(2), pp. 447-468;
- Sachs, Jeffrey D. and Andrew M. Warner (1995), *Natural Resource Abundance and Economic Growth*, NBER, Working Paper 5398;
- Smith, Daniel Jordan (2007), A Culture of Corruption: Everyday Deception and Popular Discontent in Nigeria, Princeton University Press;
- Vicente, Pedro C. (2006), *Does Oil Corrupt? Evidence from a Natural Experiment in West Africa*, Journal of Development Economics, forthcoming;
- Vicente, Pedro C. (2007a), *Is Vote Buying Effective? Evidence from a Field Experiment in West Africa*, Oxford University and BREAD, Working Paper;
- Vicente, Pedro C. (2007b), Corrupted Scholarships, Oxford University, Working Paper;
- Wantchekon, Leonard (2003), Clientelism and Voting Behavior: Evidence from a Field Experiment in Benin, World Politics, 55, pp. 399-422.

Appendix 1: Fieldwork

Treatment Design

The anti-violence campaign reached 12 enumeration areas (neighborhoods or villages). As referred, this was a random choice within the chosen 6 states of Nigeria. The randomization was conducted by the authors. It was implemented by ActionAid International Nigeria (AAIN - <u>http://www.actionaid-nigeria.org/</u>) during a two-week period, February 13th to 27th, 2007, just after the baseline survey.

AAIN is Nigeria's chapter of global ActionAid, headquartered in South Africa, with total budget USD 133m in 2005, and specializing on Community Improvement and Capacity Building – it was ranked 20th worldwide in terms of 'performance' in a recent list compiled by Financial Times/Dalberg of global organizations devoted to philanthropy (above Transparency International, UNDP, or Amnesty International)²⁰.

For this campaign, AAIN worked with local state-level partner NGOs, who actually conducted the campaign activities in the field. The authors of the paper supervised operations in the field in 4 out of the 6 states involved in this campaign. Since each state was allocated a different team of campaign fieldworkers, campaign activities were fully contemporaneous in all states.

The campaign consisted of a clear message against electoral violence, embedded in the main slogan ('No to political violence! Vote against violent politicians.'). The main guidelines of the campaign were discussed with the authors, the central AAIN officers, and the local partnering NGO representatives. Actual material and activity design was undertaken with the help of a specialized firm in Abuja.

The slogan of the campaign was written in a wide range and large quantity of distributed campaign materials: t-shirts (3,000, i.e. 1 for every 2 households in treated locations on average), caps (3,000), hijabs for Muslim women (1,000), leaflets (5,000), posters (3,000), and stickers (3,000) – images for the leaflets/posters and stickers are displayed in Appendix 2, Figure A1. Note that these means of campaigning are the ones primarily chosen by politicians in Nigeria to

²⁰ See the Financial Times, July 5th, 2007, Report on 'Corporate Citizenship and Philanthropy'.

licitly spread awareness about their candidacies. The campaign also included road shows. These were used to help widening the coverage of the main message and the distribution of materials (particularly important in urban locations), and included jingles in Yoruba, Hausa, and Pidgin English.

The campaign was however designed to work mainly through the holding of town meetings and popular theatres. The town meetings were designed to provide an opportunity for grassroots to meet with local representatives for finding ways of counteracting politically-motivated violence – theoretically they were built to minimize the collective action problem that is associated to finding effective ways of diminishing conflict at the local level. Popular theatres were based on the same script for all states (featuring one good and one bad politician, with the bad one instilling violent intimidation), and were designed to target youths (usually the ones providing labor for violent activities) and other audiences (e.g. women) not as easy to recruit for town meetings²¹. The campaigners were instructed to primarily target panel respondents (i.e. baseline subjects²²), not only in terms of distribution of materials, but also in terms of invitations to attending the town meetings and the popular theatres.

The campaign broadly happened in orderly terms, without substantial obstacles, in a timely manner. Some examples of the worst problems confronted are given in the following passages of state-level reports:

Any campaign in Warri cannot be carried out without police involvement. As such the Divisional Police Officer was contacted; five police officers were allotted for the campaign at Ogiame Primary School. [...] The campaign went on smoothly. – Delta state.

'The last day of the campaign in Rukpakwolusi witnessed the storming of the community by militants of the Niger Delta People Volunteers Force shooting sporadically into the air in a convoy of vehicles at the exact location where the community had gathered to witness the campaign drama. When the militants

²¹ A report of the campaign, including systematic photos and films for each state's campaign activities is available from <u>http://www.iig.ox.ac.uk/iig/research/09-political-violence-nigeria/</u>. These include a thorough depiction of the all campaign materials and activities.
²² At least one campaign representative accompanied the survey team during the pre-cleation survey.

 $^{^{22}}$ At least one campaign representative accompanied the survey team during the pre-election survey, primarily for site identification, and respondent addresses were shared with the campaigners at that stage.

alighted from their vehicles, some of them recognized the presence of Commonwealth of Niger Delta Youths leadership at the venue and actually did not harass anybody, but requested that campaign materials should be distributed to them. That was complied on the directive of the state project coordinator.' Rivers state.

'The thug nature of Kasuwan-nama community [part of North Jos treatment enumeration area] members disturbed the fluid process the campaign anticipated. The most scary part was the sharing of campaign materials which resulted into a rowdy situation until we were forced to exit the scene.' – Plateau state.

Sampling

Our field experiment included 24 locations/enumeration areas. These were chosen within Afrobarometer's (<u>http://www.afrobarometer.org/</u>) representative sample of all 36 states of Nigeria that served in their 2007 pre-election survey (contemporaneous to ours). Their sample choice used census data (i.e. weighted by the distribution of households) and corresponding enumeration areas (all averaging approximately 500 households). It included 301 enumeration areas.

Our sampling has a non-random component as we chose 2 states in each of the three main regions of the country (Southwestern, Southeastern, and Northern), by looking at the recent history of politically-motivated violence. Namely we used reports by Human Rights Watch²³, ActionAid International, and other independent sources. This process led to choosing Lagos and Oyo (Southwestern), Delta and Rivers (Southeastern), and Kaduna and Plateau (Northern). This option revealed our emphasis on studying violence over studying the 'country' (which would be achieved under a representative choice of states), while keeping the basic, traditional diversity of the country - very much borrowed from the conventional ethnic predominance of Yoruba in the southwest, Igbo in the southeast, and Fulani/Hausa in the north.

The remaining stage of the sampling process was purely random. We began by randomly choosing 2 'treated' enumeration areas in each of the 6 states from Afrobarometer's list. Then we chose 'control' areas by selecting the closest locations from each of the 'treated' that were in

²³ See for instance Human Rights Watch, 'Testing Democracy: Political Violence in Nigeria', April 2003, 'Nigeria's 2003 Elections: the Unacknowledged Violence', June 2004.
Afrobarometer's list and that were of the same type in terms of the classification 'large urban', 'small urban', and 'rural' (stemming from the census). This process led to choosing the areas shown in Map A1, Appendix 2.

The data-gathering fieldwork was performed jointly with Afrobarometer and their long-standing Nigerian partner Practical Sampling International (PSI) for the baseline survey. At that time, households within a census area were chosen randomly using Afrobarometer's standard techniques (e.g. by choosing the nth house). Then the conditions for sampling within a household were: 18 years of age or higher, residence in the corresponding enumeration area. In the baseline we interviewed 1200 respondents (50 per enumeration area) in the period January 20th to February 3rd, 2007.

In the post-election survey, we were on the field through PSI, though not involving Afrobarometer's explicit collaboration. At that time we sought to visit the same respondents that were included in the sample for the baseline – we could re-survey 1149 (96%) of those subjects in the period May 22nd to June 5th, 2007. During the same time frame, we gathered a fresh sample of 300 respondents in treated enumeration areas (25 per enumeration area), what we will refer to as the treatment post-election oversample. The only additional requirement for this group's sampling procedure (apart from the referred standard rules) was that they were 'not directly approached' by the campaign team. This way we made sure that this group could serve as a control group within treatment areas for direct campaigning.

The authors supervised the whole fieldwork, and were in-field during training and piloting of the surveys.

Questionnaire Design

On questionnaire design, while the baseline was jointly designed with Afrobarometer, therefore including some modules that are not related with our field experiment (e.g. public opinion questions), the post-election instruments (panel and oversample) were fully targeting this experiment.²⁴ This setting implied that in the post-election surveys we could include a larger number of questions related to our measurements of interest (e.g. including extended demographics). The basic structure of questionnaires was however untouched across different

²⁴ All questionnaires are available upon request to the authors.

instruments, including by order, demographics, the different measures of violence, information about elections, and self-reported electoral preferences/behavior.

Note that most prominently on violence, the majority of the questions featured a subjective scale. These scales (usually) had an odd number of options, were approached verbally using qualifiers (not numbers), and were read in a stepwise manner. For instance, the question 'How secure against violence originated by politicians has been your neighborhood or village?' embedded 7 possible answers ranging from 'Extremely insecure' to 'Extremely secure', with middle answers 'neither insecure nor secure', and 'somewhat', 'very' insecure/secure. These options were approached first by referring to a 3-point scale: 'insecure', 'neither insecure nor secure', and 'secure'. Then, depending on the answer, the enumerator could deepen one of the sides. This careful and balanced way of reading scales gives us some assurance that original question-scales were perceived linearly by respondents.

The Postcard Variable

One of the measurements that only took place during the post-election period was the one relating to our 'postcard variable'. During the interview in the post-election surveys (panel and oversample), we provided the respondents with a pre-stamped (ready-to-mail) postcard – the main side of the postcard is shown in Appendix 2, Figure A2 (it was addressed to PSI in Lagos on the other side and did not require the sender to identify him/herself). Enumerators explained to respondents that sending the card would contribute to having the sponsors of this project highlight the concern for the problem of political violence in the respondent's state. Each postcard was numbered and we could therefore identify the sender. We interpret the sending of this postcard as an incentive-compatible measure of empowerment to counteract violence, as it embedded an individual cost of taking the postcard to a post office²⁵.

²⁵ We nevertheless control for obvious determinants of postcard sending like income and distance to post office.

Appendix 2: Field Materials and Sample

Figure A1: Leaflet/Poster (above) and Sticker (below)





Figure A2: Postcard



Map A1: Sampled Enumeration Areas





- 1. Alimosho Akwonjo LU
- 2. Alimosho Ikotun LU 3. Lagos Mainland – Ebute Met. LU
- 4. Lagos Island Lagos Island LU

13. Jos North - Jos (78) LU 14. Jos North – Jos (77) LU

15. Quan-Pan – Piya R 16. Quan-Pan – Pandam R

- 21. Andoni Agama R 22. Eleme – Sime-Tai R
- 23. Obio/Akpor Rukpakwolusi R
- 24. Gokana Nugbe-Yeghe R

Appendix 3: Regression Tables

| | | | | | Ex | perimental S | ample - 6 Sta | ites | | | |
|---------------------------------|---------|---------|---------------|----------------|-----------------------|--------------|------------------------|---------|------------|----------------|---------------|
| | | Eleo | ctoral Violen | ce (post-elect | ion) | | raud - (post- tion) | ۲ | ote Buying | (post-election | 1) |
| Dependent Variable> | | viol | ence | | [pdp/opp iolence)] | fra | ud | v | b | vb*[pdp/opj | p (doing vb)] |
| | | | | | | 0 | LS | | | • | |
| urban | coef | 0.06 | 0.42 | -0.25 | 1.1 | 0.45 | 0.04 | 0.43 | 0.09 | 2.27 | 0.93 |
| urban | std err | 0.2 | 0.11*** | 1.09 | 0.48** | 0.26 | 0.21 | 0.28 | 0.21 | 2.03 | 1.98 |
| aha(ada ana) in 2002 | coef | 0.75 | 0.74 | 1.17 | 1.39 | -0.39 | -1.58 | -0.41 | -1.23 | -5.42 | -9.42 |
| abs(pdp-opp) in 2003 | std err | 0.28** | 0.12*** | 0.99 | 0.66** | 0.52 | 0.39*** | 0.57 | 0.34*** | 2.61** | 3.01*** |
| n de in south ant | coef | -0.12 | 0.28 | -2.94 | -1.85 | 0.19 | 0.46 | 0.21 | 0.48 | 14.16 | 14.99 |
| pdp incumbent | std err | 0.23 | 0.2 | 0.70*** | 0.73** | 0.36 | 0.24* | 0.37 | 0.33 | 2.28*** | 1.89*** |
| | coef | 2.47 | 2.23 | 6.34 | 3.65 | 2.08 | 2.64 | 2.09 | 2.2 | 2 | 3.63 |
| Constant | std err | 0.29*** | 0.24*** | 1.21*** | 1.05*** | 0.29*** | 0.38*** | 0.32*** | 0.32*** | 2.05 | 1.72** |
| Regional Dummies | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Infrastructure/Demographic Cont | ols | No | Yes | No | Yes | No | Yes | No | Yes | No | Yes |
| Number of Observations | | 1434 | 1411 | 1434 | 1407 | 1384 | 1192 | 1366 | 1181 | 1366 | 1181 |
| Adjusted R-squared (OLS) | | 0.11 | 0.2 | 0.61 | 0.66 | 0.08 | 0.15 | 0.09 | 0.17 | 0.67 | 0.7 |

Table A1: Strategic Determinants of Violence, Electoral Fraud, and Vote Buying in the 2007 Elections

Note: Standard errors reported; these are corrected by clustering at the location (census area) level. * significant at 10%; *** significant at 5%; *** significant at 1%.

| | | | Afrobaro | meter Natio | nal Sample - | 37 States | |
|-------------------------------|---------|---------|-----------------------|-------------|----------------------|-----------|--------------------|
| | | | olence - pre- tion | | 'raud - pre- tion | | ing - pre- tion |
| Dependent Variable> | | viol | ence | fra | ud | v | b |
| | | | | 0 | LS | | |
| urban | coef | -0.05 | -0.08 | 0.05 | 0.08 | -0.04 | 0 |
| urban | std err | 0.05 | 0.05 | 0.06 | 0.06 | 0.06 | 0.07 |
| aba(ada ana) | coef | 0.08 | 0.19 | -0.04 | -0.2 | 0.02 | -0.18 |
| abs(pdp-opp) | std err | 0.07 | 0.07*** | 0.11 | 0.12* | 0.1 | 0.11* |
| | coef | -0.44 | -0.47 | 1.07 | 1.12 | 1.15 | 1.17 |
| pdp incumbent | std err | 0.18** | 0.20** | 0.32*** | 0.41*** | 0.30*** | 0.34** |
| Constant | coef | 3.33 | 3.02 | 1.84 | 2.97 | 1.9 | 2.64 |
| Constant | std err | 0.15*** | 0.22*** | 0.11*** | 0.21*** | 0.13*** | 0.22** |
| Regional Dummies | | Yes | Yes | Yes | Yes | Yes | Yes |
| Infrastructure/Demographic Co | ntrols | No | Yes | No | Yes | No | Yes |
| Number of Observations | | 2292 | 1833 | 2070 | 1710 | 2057 | 1577 |
| Adjusted R-squared (OLS) | | 0.08 | 0.11 | 0.14 | 0.18 | 0.16 | 0.23 |

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

| | | Control | Treatment | Difference | Number of Observations |
|-----------------------------|--|---------|-----------|---------------------|---------------------------|
| | female | 0.5 | 0.5 | 0 | 1500 |
| | age | 32.95 | 32.69 | 0 -0.26 1.01 | 1497 |
| | household size | 6.43 | 6.46 | 0.03 0.74 | 1500 |
| Basic Demographics | single | 0.38 | 0.42 | 0.04 0.05 | 1449 |
| | married | 0.58 | 0.55 | -0.03 0.04 | 1449 |
| | schooling (0-9) | 4.31 | 4.83 | 0.52 0.34 | 1500 |
| | yoruba | 0.32 | 0.28 | -0.04 0.17 | 1500 |
| | hausa | 0.16 | 0.1 | -0.06 0.11 | 1500 |
| | igbo | 0.07 | 0.16 | 0.09 0.09 | 1500 |
| Ethnicity and Religion | christian | 0.62 | 0.74 | 0.12 0.13 | 1499 |
| | muslim | 0.34 | 0.25 | -0.09 0.13 | 1499 |
| | religious intensity (1-6) | 4.76 | 5.07 | 0.31 0.2 | 1485 |
| | job stability (0-3) | 1.36 | 1.41 | 0.05 0.16 | 1500 |
| | agriculture | 0.16 | 0.12 | -0.04 0.07 | 1500 |
| | public official | 0.07 | 0.07 | 0 0.01 | 1500 |
| Ormertin | industry/services: trader | 0.13 | 0.14 | 0.01 0.03 | 1500 |
| Occupation | industry/services: artisan | 0.11 | 0.13 | 0.02 0.03 | 1500 |
| | industry/services: other | 0.11 | 0.15 | 0.04 0.04 | 1500 |
| | student | 0.22 | 0.22 | 0 0.04 | 1500 |
| | housework | 0.12 | 0.09 | -0.03 0.04 | 1500 |
| | house | 0.61 | 0.58 | -0.03 0.11 | 1496 |
| | land | 0.53 | 0.56 | 0.03 0.12 | 1494 |
| | cattle | 0.33 | 0.37 | 0.04 | 1495 |
| | tv | 0.63 | 0.72 | 0.09 0.11 | 1497 |
| Property and Expenditure | radio | 0.89 | 0.93 | 0.04 | 1499 |
| | fridge | 0.29 | 0.37 | 0.08 0.11 | 1496 |
| | land phone | 0.1 | 0.1 | 0 0.05 | 1496 |
| | cell phone | 0.51 | 0.58 | 0.07 0.12 | 1496 |
| | household expenditure (naira/month) | 19001 | 22869 | 3867 4759 | 1262 |

 Table A2: Differences across Treatment and Control Areas - Demographics (Panel plus Oversample)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

* significant at 10%; ** significant at 5%; *** significant at 1%. These results come from OLS regressions.

| | | Control | Treatment | Difference | Number of Observation |
|--------------------|---|---------|-----------|----------------------|--------------------------|
| | free & fair 2007 election - general (1-4) | 3.24 | 3.00 | -0.24 0.15 | 961 |
| | free & fair 2007 election - violence (1-4) | 2.98 | 2.87 | -0.10 0.17 | 977 |
| | conflict within family (0-4) | 0.61 | 0.6 | -0.01 0.07 | 1194 |
| | conflict within community (0-4) | 1.18 | 1.29 | 0.11 0.18 | 1184 |
| | political intimidation (0-3) | 0.09 | 0.1 | 0.01 0.03 | 1089 |
| | influence of assassinations (1-7) | 3.35 | 3.05 | -0.3 0.33 | 1125 |
| | politicians advocating violence (1- 7) | 2.9 | 3.29 | 0.39 0.29 | 1128 |
| | thugs' violent activities (1-7) | 3.03 | 2.77 | -0.26 0.44 | 1140 |
| | security (1-7) | 4.53 | 4.55 | 0.44 0.02 0.28 | 1144 |
| | support for 'do-or-die affair' (1-7) | 3.2 | 3.32 | 0.28 0.12 0.35 | 1135 |
| | standing against violence (1-7) | 4.85 | 4.97 | 0.12 | 1136 |
| Violence | empowerment against violence (1- 7) | 4.25 | 4.81 | 0.26 0.56 | 1116 |
| | knowledge of ways to counteract | 4.75 | 4.62 | 0.31* -0.13 | 1133 |
| | violence (1-7) 'violence is justified' (1-5) | 2.08 | 2.05 | 0.3 -0.03 | 1181 |
| | thefts public place (perception) (1- | 2.88 | 2.59 | 0.14 -0.29 | 1142 |
| | 7) thefts public place (experience) (1- | 1.17 | 1.26 | 0.38 0.09 | 1193 |
| | 4) vandalism (perception) (1-7) | 2.42 | 2.19 | 0.06 -0.23 | 1130 |
| | vandalism (experience) (1-4) | 1.09 | 1.05 | 0.36 -0.04 | 1137 |
| | physical intimidation (perception) | 2.34 | 2.3 | 0.03 -0.04 | 1145 |
| | (1-7) physical intimidation (experience) | 1.14 | 1.09 | 0.37 -0.05 | 1195 |
| | (1-4) problems with police (1-7) | 2.76 | 2.56 | 0.04 -0.2 | 1122 |
| | | 2.70 | 2.30 | 0.32 0.01 | 57 |
| | intensity (journals) (1-5) | | | 0.24 | <u> </u> |
| | turnout 2003 | 0.7 | 0.7 | 0.06 0.03 | 1174 |
| | pdp 2003 presidential | 0.45 | 0.48 | 0.09 | 1149 |
| | anpp 2003 presidential | 0.16 | 0.08 | 0.08 | 1149 |
| Electoral Behavior | * | 0.03 | 0.05 | 0.02 | 1149 |
| | pdp 2003 governor | 0.45 | 0.46 | 0.01 | 1149 |
| | anpp 2003 governor | 0.13 | 0.09 | 0.06 | 1149 |
| | ac 2003 governor | 0.03 | 0.03 | 0.02 | 1149 |
| Panel Attrition | panel re-surveying | 0.97 | 0.95 | -0.02 0.01 | 1200 |

Table A2 (continued): Differences across Treatment and Control Areas - Panel (Baseline Violence and Political Preferences, Attrition)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

* significant at 10%; ** significant at 5%; *** significant at 1%. These results come from OLS regressions.

| | | | | | | Do | litical Enco | dom Cono | | | | | |
|--------------------------|---|--|--|--|---|--|--|---|---|---|---|--|---|
| | | | | | | PO | intical r ree | uom - Gene | rai | | | | |
| Dependent Variable | > | Change of Freedom to Think | | Change of Freedom to Join a Party | | Change of Freedom to Vote Freely | | Change of Freedom from Crime and Insecurity | | | | Free & Fair 200 Elections - Viole | |
| | | | | | | | 0 | LS | | | | | |
| timo | coef | | | | | | | | | -0.8 | -0.92 | -0.58 | -0.7 |
| ume | std err | | | | | | | | | 0.23*** | 0.19*** | 0.24** | 0.20*** |
| troot | coef | 0.37 | 0.42 | 0.33 | 0.37 | 0.36 | 0.41 | 0.43 | 0.47 | -0.25 | -0.24 | -0.11 | -0.08 |
| licai | std err | 0.11*** | 0.10*** | 0.10*** | 0.09*** | 0.14** | 0.12*** | 0.15*** | 0.13*** | 0.17 | 0.15 | 0.16 | 0.15 |
| time*treat | coef | | | | | | | | | 0.49 | 0.63 | 0.46 | 0.55 |
| time+treat | std err | | | | | | | | | 0.26* | 0.22*** | 0.26* | 0.22** |
| Constant | coef | 3.49 | 3.9 | 3.72 | 3.79 | 3.6 | 4.11 | 2.77 | 3.38 | 3.24 | 3.31 | 2.83 | 2.83 |
| Constant | std err | 0.14*** | 0.21*** | 0.12*** | 0.17*** | 0.17*** | 0.30*** | 0.19*** | 0.26*** | 0.14*** | 0.32*** | 0.18*** | 0.34*** |
| al Fixed Effects/Dummies | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| mographic Controls | | No | Yes | No | Yes | No | Yes | No | Yes | No | Yes | No | Yes |
| mber of Observations | | 1135 | 960 | 1132 | 960 | 1132 | 962 | 1133 | 960 | 2065 | 1710 | 2081 | 1728 |
| sted R-squared (OLS) | | 0.09 | 0.13 | 0.13 | 0.16 | 0.12 | 0.16 | 0.11 | 0.16 | 0.08 | 0.14 | 0.07 | 0.13 |
| 2 | Dependent Variable time treat time*treat Constant al Fixed Effects/Dummies emographic Controls mber of Observations isted R-squared (OLS) | time std err treat coef std err time*treat coef std err Constant coef std err coef std err tid err al Fixed Effects/Dummies emographic Controls mber of Observations | $\frac{coef}{std \ err}$ \frac{coef} | Coef Ime time coef | b b to Think to Join to Think to Join to Think to Join time coef c | Coef Image: coef <th< td=""><td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td><td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td><td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td><td>Dependent Variable> to Think to Join a Party to Vote Freely from Crime and Insecurity Dependent Variable> coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party time*treat coef doin a Party <th< td=""><td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td><td>Preedom to ThinkChange of Freedom to Join a PartyChange of Freedom to Vote FreelyChange of Freedom from Crime and InsecurityFreedom from Crime and Insecuritytimecoefcoefcococososososotime*treatcoef0.370.420.330.370.360.410.430.47-0.25-0.24time*treatcoef0.11***0.10***0.10***0.09***0.14***0.12***0.12***0.12***0.12***0.15***0.13***0.13***0.170.15***time*treatcoef3.493.93.723.793.64.112.773.383.243.31time*treatcoef3.493.93.723.793.64.112.773.383.243.31al Fixed Effects/Dummies<!--</td--><td>Dependent Variable>Change of Freedom to Join a PartyChange of Freedom to Vote FreelyGrange of Freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and Insecuritytimecoefcoef0.370.420.330.370.360.410.430.47-0.25-0.24-0.11time*coef0.370.420.10***0.09***0.14***0.12***0.16***0.12***0.13***0.170.150.16time*coef3.493.93.723.793.64.11</br></br></br></br></br></br></td></td></th<></td></th<> | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | $ \begin{array}{c c c c c c c c c c c c c c c c c c c $ | $ \begin{array}{c c c c c c c c c c c c c c c c c c c $ | Dependent Variable> to Think to Join a Party to Vote Freely from Crime and Insecurity Dependent Variable> coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party doin a Party to Vote Freely from Crime and Insecurity time coef doin a Party time*treat coef doin a Party doin a Party doin a Party doin a Party doin a Party <th< td=""><td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td><td>Preedom to ThinkChange of Freedom to Join a PartyChange of Freedom to Vote FreelyChange of Freedom from Crime and InsecurityFreedom from Crime and Insecuritytimecoefcoefcococososososotime*treatcoef0.370.420.330.370.360.410.430.47-0.25-0.24time*treatcoef0.11***0.10***0.10***0.09***0.14***0.12***0.12***0.12***0.12***0.15***0.13***0.13***0.170.15***time*treatcoef3.493.93.723.793.64.112.773.383.243.31time*treatcoef3.493.93.723.793.64.112.773.383.243.31al Fixed Effects/Dummies<!--</td--><td>Dependent Variable>Change of Freedom to Join a PartyChange of Freedom to Vote FreelyGrange of Freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and Insecuritytimecoefcoef0.370.420.330.370.360.410.430.47-0.25-0.24-0.11time*coef0.370.420.10***0.09***0.14***0.12***0.16***0.12***0.13***0.170.150.16time*coef3.493.93.723.793.64.11</br></br></br></br></br></br></td></td></th<> | $ \begin{array}{c c c c c c c c c c c c c c c c c c c $ | Preedom to ThinkChange of Freedom to Join a PartyChange of Freedom to Vote FreelyChange of Freedom from Crime and InsecurityFreedom from Crime and Insecuritytimecoefcoefcococososososotime*treatcoef0.370.420.330.370.360.410.430.47-0.25-0.24time*treatcoef0.11***0.10***0.10***0.09***0.14***0.12***0.12***0.12***0.12***0.15***0.13***0.13***0.170.15***time*treatcoef3.493.93.723.793.64.112.773.383.243.31time*treatcoef3.493.93.723.793.64.112.773.383.243.31al Fixed Effects/Dummies </td <td>Dependent Variable>Change of Freedom to Join a PartyChange of Freedom to Vote FreelyGrange of Freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom from Crime and Insecuritytimecoefcoef0.370.420.330.370.360.410.430.47-0.25-0.24-0.11time*coef0.370.420.10***0.09***0.14***0.12***0.16***0.12***0.13***0.170.150.16time*coef3.493.93.723.793.64.11</br></br></br></br></br></br></td> | Dependent Variable>Change of Freedom to Join a PartyChange of Freedom to Vote FreelyGrange of Freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom freedom from Crime and InsecurityFreedom from Crime and InsecurityFreedom |

Table A3: Regressions of Violence Outcomes (on Treatment)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level. * significant at 10%; ** significant at 5%; *** significant at 1%.

| | | | | | E | lectoral Vio | olence - Loc | al | | |
|------------------|--------------------------|---------|---------|------------------|---------|-----------------------|--------------|-------------|---------|---------------------|
| | Dependent Variable | > | | t within nily | | ithin Local nunity | Political Ir | itimidation | 0 | f Political ence |
| | - | | | | | 0 | LS | | | |
| | time | coef | -0.43 | -0.41 | -0.44 | -0.34 | 0.17 | 0.19 | | |
| | ume | std err | 0.05*** | 0.05*** | 0.15*** | 0.14** | 0.06*** | 0.06*** | | |
| Г | treat | coef | -0.01 | -0.04 | 0.11 | 0.1 | 0.01 | 0 | -0.51 | -0.57 |
| Main Explanatory | treat | std err | 0.06 | 0.06 | 0.11 | 0.13 | 0.03 | 0.03 | 0.29* | 0.29* |
| Variables | time*treat | coef | -0.06 | -0.07 | -0.42 | -0.48 | -0.16 | -0.15 | | |
| | time-treat | std err | 0.07 | 0.07 | 0.22* | 0.21** | 0.07** | 0.07** | | |
| | Constant | coef | 0.61 | 0.82 | 1.47 | 1.55 | 0.07 | -0.1 | 3.4 | 2.07 |
| | Constant | std err | 0.05*** | 0.12*** | 0.18*** | 0.27*** | 0.04* | 0.07 | 0.55*** | 0.63*** |
| Regiona | al Fixed Effects/Dummies | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| De | mographic Controls | | No | Yes | No | Yes | No | Yes | No | Yes |
| Nur | nber of Observations | | 2332 | 1922 | 2314 | 1906 | 2168 | 1827 | 1064 | 881 |
| Adju | sted R-squared (OLS) | | 0.12 | 0.12 | 0.16 | 0.16 | 0.03 | 0.05 | 0.12 | 0.16 |

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

| Table A3 (continued): Regressions of Violence Outcomes (on Treatment) | Table A3 | 3 (continued): | Regressions | of Violence | Outcomes | (on Treatment) |
|---|----------|----------------|-------------|-------------|----------|----------------|
|---|----------|----------------|-------------|-------------|----------|----------------|

| | | | | | Electoral V | Violence - L | ocal and fro | om the Top | | |
|------------------|-----------------------|---------|---------|---------------------|-------------|----------------------|--------------|------------|---------|---------|
| | Dependent Variable | > | | ence of inations | | icians g Violence | Gang A | Activity | Seci | ırity |
| | | | | | | 0 | LS | | | |
| | time | coef | 0.01 | 0.03 | -0.04 | 0.03 | -0.12 | -0.08 | 0.12 | 0.07 |
| | time | std err | 0.13 | 0.14 | 0.13 | 0.13 | 0.17 | 0.17 | 0.12 | 0.14 |
| | treat | coef | -0.29 | -0.29 | 0.39 | 0.57 | -0.26 | -0.18 | 0.03 | 0.01 |
| Main Explanatory | treat | std err | 0.16* | 0.14** | 0.20* | 0.17*** | 0.28 | 0.24 | 0.24 | 0.21 |
| Variables | time*treat | coef | -0.34 | -0.38 | -0.36 | -0.48 | -0.44 | -0.5 | 0.46 | 0.57 |
| | time-treat | std err | 0.20* | 0.22* | 0.16** | 0.19** | 0.21* | 0.23** | 0.18** | 0.20*** |
| | Constant | coef | 3.88 | 2.89 | 2.8 | 2.01 | 3.72 | 1.92 | 4.4 | 5.12 |
| | Constant | std err | 0.20*** | 0.36*** | 0.42*** | 0.56*** | 0.57*** | 0.55*** | 0.33*** | 0.39*** |
| R | egional Fixed Effects | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| D | emographic Controls | | No | Yes | No | Yes | No | Yes | No | Yes |
| Nu | mber of Observations | | 2252 | 1946 | 2257 | 1907 | 2280 | 1919 | 2287 | 1926 |
| Adjı | usted R-squared (OLS) | | 0.13 | 0.15 | 0.1 | 0.14 | 0.2 | 0.25 | 0.06 | 0.1 |

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

* significant at 10%; ** significant at 5%; *** significant at 1%.

| | | | | | | E | mpowerme | nt Against 1 | Political Vie | olence - Loca | al and fron | 1 the Botton | n | | | |
|------------------|---------------------------|------------|--------------------|-----------------------|---------|-------------------|----------|---------------------|---------------|--------------------------------|-------------|--------------|-------------|--------|----------------|-------------------|
| | Dependent Variable | > | Support 1 Die A | for 'Do or \ffair' | | g against ence | - | verment Violence | to Cou | ge of Ways nteract lence | Postc | ard (Vote f | or Dissemin | ation) | 'Viole Just | ence is ified' |
| | | | | | | | | | | | | Pr | obit | | | LS |
| | | | | | | | | | | | | ME | | ME | 0 | 1.5 |
| | timo | coef | -0.02 | -0.01 | 0.5 | 0.45 | 0.4 | 0.38 | 0.55 | 0.5 | | | | | -0.27 | -0.27 |
| | time | std err | 0.17 | 0.18 | 0.10*** | 0.10*** | 0.12*** | 0.11*** | 0.12*** | 0.12*** | | | | | 0.13* | 0.15* |
| | twoot | coef | 0.12 | 0.22 | 0.13 | 0.02 | 0.55 | 0.55 | -0.14 | -0.12 | 0.17 | 0.06 | 0.38 | 0.14 | -0.03 | 0 |
| Main Explanatory | treat | std err | 0.28 | 0.25 | 0.22 | 0.21 | 0.23** | 0.20** | 0.15 | 0.16 | 0.21 | 0.00 | 0.23* | 0.14 | 0.13 | 0.12 |
| Variables | time*treat | coef | -0.65 | -0.73 | 0.27 | 0.43 | 0.3 | 0.41 | 0.29 | 0.47 | | | | | -0.11 | -0.08 |
| | time-treat | std err | 0.22*** | 0.24*** | 0.16* | 0.15*** | 0.15* | 0.14*** | 0.16* | 0.16*** | | | | | 0.21 | 0.23 |
| ſ | Constant | coef | 3.01 | 2.84 | 4.89 | 5.04 | 4.44 | 4.38 | 5.17 | 5.32 | -0.23 | | -0.72 | | 2.14 | 3.05 |
| | Constant | std err | 0.34*** | 0.46*** | 0.15*** | 0.36*** | 0.15*** | 0.50*** | 0.11*** | 0.36*** | 0.17 | | 0.27*** | | 0.10*** | 0.25*** |
| Region | al Fixed Effects/Dummies | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Y | 'es | Y | es | Yes | Yes |
| De | emographic Controls | | No | Yes | No | Yes | No | Yes | No | Yes | Ν | No | Y | es | No | Yes |
| Nu | mber of Observations | | 2271 | 1922 | 2272 | 1917 | 2233 | 1882 | 2267 | 1913 | 11 | 149 | 9 | 73 | 2322 | 1924 |
| Adjusted R- | squared (OLS) Pseudo R2 (| P) | 0.08 | 0.12 | 0.09 | 0.13 | 0.17 | 0.2 | 0.21 | 0.22 | 0. | .02 | 0. | 05 | 0.03 | 0.05 |

| | | | | | | | | Crime | Perception | s and Expe | rience) | | | | | |
|------------------|----------------------|---------|---------------------|-----------------------|---------|-----------------------|---------|------------------|------------|------------------|-------------|------------------------|----------------------|-----------------------|---------|-----------------|
| | Dependent Variable - | > | Thefts Pu (Perce | blic Place eption) | | blic Place rience) | | alism eption) | Vand | alism rience) | Psysical In | ntimidation eption) | Psysical In (Expe | timidation rience) | | ms with lice |
| | | | | | | | | | OI | S | | | | | | |
| | time | coef | -0.19 | -0.19 | 0.04 | 0.07 | -0.06 | -0.06 | -0.03 | -0.02 | -0.05 | -0.05 | -0.04 | -0.04 | -0.2 | -0.19 |
| | time | std err | 0.10* | 0.10* | 0.07 | 0.09 | 0.1 | 0.12 | 0.01*** | 0.01* | 0.12 | 0.12 | 0.05 | 0.05 | 0.10* | 0.11 |
| | twoot | coef | -0.3 | -0.28 | 0.09 | 0.12 | -0.23 | -0.17 | -0.04 | -0.05 | -0.04 | -0.02 | -0.05 | -0.05 | -0.2 | -0.07 |
| lain Explanatory | treat | std err | 0.31 | 0.27 | 0.05* | 0.06** | 0.3 | 0.27 | 0.03 | 0.03* | 0.27 | 0.24 | 0.03 | 0.03 | 0.26 | 0.22 |
| Variables | time*treat | coef | -0.33 | -0.33 | -0.21 | -0.26 | -0.25 | -0.29 | 0 | 0 | -0.45 | -0.45 | 0.04 | 0.04 | -0.23 | -0.28 |
| | time treat | std err | 0.16** | 0.16** | 0.09** | 0.10** | 0.13* | 0.16* | 0.02 | 0.02 | 0.15*** | 0.16*** | 0.05 | 0.05 | 0.13* | 0.16* |
| | Constant | coef | 3.63 | 2.12 | 1.26 | 1.27 | 3.06 | 2.23 | 1.15 | 1.17 | 2.47 | 1.96 | 1.12 | 1.13 | 3.2 | 2.17 |
| | Constant | std err | 0.56*** | 0.59*** | 0.04*** | 0.13*** | 0.65*** | 0.34*** | 0.08*** | 0.08*** | 0.54*** | 0.40*** | 0.03*** | 0.05*** | 0.45*** | 0.46*** |
| Re | gional Fixed Effects | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| De | mographic Controls | | No | Yes | No | Yes | No | Yes | No | Yes | No | Yes | No | Yes | No | Yes |
| Nur | mber of Observations | | 2285 | 2257 | 2339 | 1925 | 2258 | 1897 | 2279 | 1920 | 2290 | 2284 | 2341 | 2313 | 2241 | 1891 |
| Adju | sted R-squared (OLS) | | 0.12 | 0.16 | 0.04 | 0.04 | 0.1 | 0.14 | 0.01 | 0.02 | 0.11 | 0.15 | 0.01 | 0.01 | 0.09 | 0.14 |

Table A3 (continued): Regressions of Violence Outcomes (on Treatment)

| Table A4: Regressions of Viole | nce Outcomes from | Journaling (on | Treatment) |
|--------------------------------|-------------------|----------------|--------------|
| Table A4. Regressions of viole | nee Outcomes nom | Journanng (on | II cauncin() |

| | | | | Elector | al Violence (Jou | rnaling) | |
|------------------|-------------------------|---------|---------|---------|------------------|----------|--------------------------|
| | Dependent Variable | > | | Inte | ensity | | Intensity & Incidence |
| | | | | | OLS | | |
| | time | coef | | -0.03 | 0 | -0.01 | -0.05 |
| | time | std err | | 0.25 | 0.17 | 0.14 | 0.14 |
| | treatment | coef | -0.16 | 0.2 | 0.33 | 0.2 | 0.02 |
| Main Explanatory | treatment | std err | 0.23 | 0.3 | 0.17* | 0.2 | 0.22 |
| Variables | time*treatment | coef | | -0.36 | -0.4 | -0.51 | -0.08 |
| | une*treatment | std err | | 0.37 | 0.28 | 0.29* | 0.42 |
| | Genetert | coef | 2.66 | 2.69 | 2.37 | 2.65 | 2.84 |
| | Constant | std err | 0.10*** | 0.25*** | 0.18*** | 0.33*** | 0.34*** |
| | Regional Dummies | | No | No | Yes | Yes | Yes |
| | Location Controls | | No | No | No | Yes | Yes |
| | Number of Observations | | 74 | 131 | 131 | 131 | 131 |
| Α | djusted R-squared (OLS) | | 0 | 0 | 0.07 | 0.06 | 0 |

| | Dependent Variable | > | | Turno | ut in 2007 I | Round of E | lections | |
|------------------|--------------------------------|---------|---------|--------|--------------|------------|----------|-------|
| | | | | Pr | obit | | IVP | robit |
| | | | | ME | | ME | | ME |
| | time | coef | 0.44 | 0.14 | 0.31 | 0.09 | | |
| | ume | std err | 0.14*** | 0.14 | 0.13** | 0.09 | | |
| | treat | coef | -0.16 | -0.05 | -0.21 | -0.06 | | |
| | ticat | std err | 0.13 | -0.05 | 0.14 | -0.00 | | |
| Main Explanatory | time*treat | coef | 0.11 | - 0.04 | 0.33 | 0.09 | | |
| Variables | | std err | 0.19 | 0.04 | 0.20* | 0.07 | | |
| | secure (inst. treat) | coef | | | | | 0.37 | 0.1 |
| | secure (mst. treat) | std err | | | | | 0.22* | 0.1 |
| | Constant | coef | 0.3 | | -0.46 | | | |
| | Constant | std err | 0.15** | | 0.28 | | | |
| Region | Regional Fixed Effects/Dummies | | | es | Y | es | Y | es |
| Demog | raphic/Political Controls | | Ň | 0 | Y | es | Y | es |
| Nu | mber of Observations | | 23 | 21 | 19 | 27 | 9′ | 77 |
| Pseu | ido R2 (P) Chi2 (IV-P) | | 0. | 05 | 0. | 09 | 198 | 3.76 |

Table A5: Regressions of Voting Behavior (Turnout)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

| | | | | | | | Voting i | n Presiden | tial Election | ns 2007 | | | | |
|-----------------|--------------------------|---------|-------|---------|----------|-------|----------|------------|---------------|---------|---------|-------|---------|-------|
| | Dependent Variable | > | | PDP (In | cumbent) | | | А | .C | | | AN | IPP | |
| | | | | | | | | Pro | bit | | | | | |
| | | | | ME | | ME | | ME | | ME | | ME | | ME |
| | time | coef | 0.01 | 0 | -0.05 | -0.02 | 0.31 | 0.05 | 0.32 | 0.04 | -0.24 | -0.03 | -0.23 | -0.0 |
| | ume | std err | 0.08 | U | 0.09 | -0.02 | 0.14** | 0.05 | 0.13** | 0.04 | 0.10** | -0.03 | 0.14 | -0.0 |
| | treat | coef | -0.23 | -0.09 | -0.23 | -0.09 | 0.44 | 0.07 | 0.42 | 0.06 | -0.52 | -0.07 | -0.49 | -0.07 |
| ain Explanatory | u cat | std err | 0.19 | -0.09 | 0.14 | -0.09 | 0.19** | 0.07 | 0.17** | 0.00 | 0.27* | -0.07 | 0.13*** | -0.0 |
| Variables | time*treat | coef | 0.07 | 0.03 | 0.18 | 0.07 | -0.39 | -0.05 | -0.38 | -0.04 | 0.1 | 0.02 | 0.09 | 0.01 |
| | time*treat | std err | 0.14 | 0.05 | 0.15 | 0.07 | 0.18** | -0.05 | 0.15** | -0.04 | 0.13 | 0.02 | 0.16 | 0.01 |
| | Constant | coef | -0.12 | | -1.2 | | -1.43 | | -1.36 | | -0.81 | | -1.88 | |
| | Constant | std err | 0.19 | | 0.35*** | | 0.20*** | | 0.29*** | | 0.28*** | | 0.47*** | |
| Reg | gional Fixed Effects | | Y | es | Y | es | Y | es | Y | es | Y | es | Y | es |
| Demogr | aphic/Political Controls | | Ν | lo | Y | es | N | 0 | Y | es | Ň | 0 | Y | es |
| Num | iber of Observations | | 23 | 49 | 19 | 30 | 23 | 49 | 22 | .14 | 23 | 49 | 19 | 50 |
| | Pseudo R2 (P) | | |) | 0. | 14 | 0.0 | 01 | 0. | 21 | 0. | 02 | 0.1 | 37 |

Table A5 (Continued): Regressions of Voting Behavior (Voting)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level. * significant at 10%; ** significant at 5%; *** significant at 1%.

| | | | | | | | | | Voting in | Gubernate | orial Election | ons 2007 | | | | | | |
|------------------|-----------------------|---------|---------|-------|---------|-------|--------|-----------|---------------|-----------|----------------|-----------|-------------|-------|---------|-----------|---------------|------|
| | Dependent Variable | > | | P | DP | |] | Incumbent | (state-level) |) | 1 | 2nd Party | state-level |) | | 3rd Party | (state-level) | |
| | | | | | | | | | | Pro | bit | | | | | | | |
| | | | | ME | | ME | | ME | | ME | | ME | | ME | | ME | | ME |
| | time | coef | 0 | 0 | -0.06 | -0.02 | 0.15 | 0.06 | 0.08 | 0.03 | -0.09 | -0.02 | -0.13 | -0.02 | -0.13 | -0.02 | 0.08 | 0.01 |
| | time | std err | 0.09 | U | 0.11 | -0.02 | 0.13 | 0.00 | 0.12 | 0.05 | 0.14 | -0.02 | 0.12 | -0.02 | 0.13 | -0.02 | 0.15 | 0.01 |
| | treat | coef | -0.3 | -0.12 | -0.29 | -0.11 | -0.27 | -0.11 | -0.3 | -0.12 | -0.26 | -0.05 | -0.3 | -0.05 | 0.02 | 0 | 0.23 | 0.03 |
| Main Explanatory | s | std err | 0.18* | -0.12 | 0.15* | -0.11 | 0.18 | -0.11 | 0.16* | -0.12 | 0.18 | -0.05 | 0.13** | -0.03 | 0.36 | U | 0.28 | 0.03 |
| Variables | | coef | 0.25 | 0.1 | 0.34 | 0.13 | 0.24 | | 0.29 | 0.11 | -0.09 | -0.02 | -0.07 | -0.01 | 0.06 | 0.01 | 0.01 | 0 |
| | time ti eat | std err | 0.12** | 0.1 | 0.13** | 0.15 | 0.15 | 0.1 | 0.13** | 0.11 | 0.16 | -0.02 | 0.16 | -0.01 | 0.18 | 0.01 | 0.18 | U |
| | Constant | coef | -0.96 | | -1 | | -0.32 | | -0.91 | | -0.86 | | -1.76 | | -2.29 | | -3.53 | |
| | Constant | std err | 0.10*** | | 0.26*** | | 0.16** | | 0.24*** | | 0.11*** | | 0.27*** | | 0.39*** | | 0.39*** | |
| Re | egional Fixed Effects | | Y | es | Y | es | Y | es | Y | es | Y | es | Y | es | Y | es | Y | es |
| De | emographic Controls | | N | 0 | Y | es | Ň | 0 | Y | es | N | 0 | Y | es | N | lo | Y | es |
| Nu | mber of Observations | | 23 | 49 | 19 | 71 | 23 | 49 | 21 | 95 | 19 | 54 | 18 | 374 | 15 | 566 | 13 | 17 |
| | Pseudo R2 (P) | | 0. | .1 | 0. | 14 | 0. | 06 | 0. | 1 | 0. | 18 | 0. | 24 | 0. | 13 | 0.2 | 24 |

| | iucu). Regi essions of | 8 | | Voting in Gubernatorial Elections 2007 | | | | | | | | | | |
|------------------|------------------------|---------|---------|--|---------|-------|-----------|----------|---------------|----------|---------|-------|-------|-------|
| | | | | | | | Voting in | Gubernat | orial Electio | ons 2007 | | | | |
| | | | | | | | I | ncumbent | (state-level) | | | | | |
| | Dependent Variable | > | La | gos | 0 | yo | De | lta | Riv | ers | Plat | teau | Kad | luna |
| | | | | | | | | Pro | bit | | | | | |
| | | | | ME | | ME | | ME | | ME | | ME | | ME |
| | time | coef | 0.92 | 0.34 | -0.43 | -0.16 | 0.33 | 0.11 | -0.06 | -0.02 | -0.08 | -0.03 | 0.31 | 0.11 |
| | time | std err | 0.16*** | 0.34 | 0.11*** | -0.10 | 0.15** | 0.11 | 0.19 | -0.02 | 0.11 | -0.05 | 0.2 | 0.11 |
| | treat | coef | 0.01 | 0 | -1.24 | -0.44 | -0.02 | -0.01 | -0.95 | -0.36 | 0.52 | 0.21 | -0.1 | -0.04 |
| Main Explanatory | | std err | 0.27 | | 0.53** | -0.44 | 0.35 | -0.01 | 0.36*** | -0.30 | 0.18*** | 0.21 | 0.27 | -0.04 |
| Variables | time*treat | coef | -0.1 | -0.04 | 0.55 | 0.21 | -0.19 | -0.07 | 0.65 | 0.25 | 0.49 | 0.19 | 0.04 | 0.02 |
| | time treat | std err | 0.16 | -0.04 | 0.22** | 0.21 | 0.16 | -0.07 | 0.26** | 0.25 | 0.18*** | 0.19 | 0.21 | 0.02 |
| | Constant | coef | -0.89 | | 0.06 | | 0.45 | | -0.46 | | -0.25 | | -0.06 | |
| | Constant | std err | 0.21*** | | 0.71 | | 0.24* | | 0.42 | | 0.09*** | | 0.43 | |
| De | mographic Controls | | Y | es | Y | es | Y | es | Y | es | Y | es | Y | es |
| Nun | nber of Observations | | 39 | 93 | 39 | 92 | 3 | 86 | 39 | 93 | 39 | 93 | 38 | 88 |
| | Pseudo R2 (P) | | 0. | 09 | 0. | 12 | 0. | 01 | 0. | 09 | 0. | 07 | 0. | 02 |

Table A5 (Continued): Regressions of Voting Behavior (Voting)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level. * significant at 10%; ** significant at 5%; *** significant at 1%.

| | | | Pı | esidential | Elections 20 | 07 | Gu | bernatorial | Elections 2 | 007 |
|----|------------------------|---------|----------|------------|--------------|------|---------|-------------|-------------|------|
| | | | | Moving | from AC | | | Moving | to PDP | |
| | Dependent Variable | > | To Abs | stention | Tol | PDP | From Al | ostention | Fron | n AC |
| | | | | | | Pr | obit | | | |
| | | | ME | | ME | | ME | | ME | |
| | treat | coef | oef 1.12 | | 0.27 | 0.07 | 1.09 | 0.25 | 0.41 | 0.07 |
| | treat | std err | 0.61* | 0.37 | 0.68 | 0.07 | 0.40*** | 0.25 | 0.46 | 0.07 |
| | Constant | coef | 0.45 | | -3.66 | | -1.76 | | -0.49 | |
| | Constant | std err | 0.86 | | 1.16*** | | 0.57*** | | 0.84 | |
| | Regional Dummies | | Y | es | Y | es | Y | es | Y | es |
| D | emographic Controls | | Y | es | Y | es | Y | es | Y | es |
| Nu | Number of Observations | | | 36 | 8 | 2 | 302 | | 22 | 28 |
| | Pseudo R2 (P) | | | 26 | 0. | 33 | 0. | 27 | 0.1 | 24 |

| | | | | Political Fre | eedom - General | | Elec | toral Violence - I | Local - General a | and from the T | Гор |
|------------------|--------------------------|---------|----------------------------------|---|--|--|------------------------------------|---------------------------------------|-------------------|----------------|------------------------------------|
| | Dependent Variable - | > | Change of Freedom to Think | Change of Freedom to Join a Party | Change of Freedom to Vote Freely | Change of Freedom from Crime and Insecurity | Change of Political Violence | Politicians Advocating Violence | Gang Activity | Security | Influence of Assassination s |
| | | | | | | | OLS | | | | |
| | timo | coef | | | | | | -0.3 | -0.46 | 0.47 | -0.17 |
| | time | std err | | | | | | 0.12** | 0.17** | 0.14*** | 0.14 |
| | treat | coef | -0.03 | 0.05 | -0.1 | 0.06 | 0.05 | 0.58 | 0.32 | -0.42 | 0.4 |
| Main Explanatory | treat | std err | 0.09 | 0.06 | 0.07 | 0.07 | 0.12 | 0.11*** | 0.14** | 0.18** | 0.17** |
| Variables | time*treat | coef | | | | | | -0.1 | -0.09 | 0.11 | -0.16 |
| | time ti cat | std err | | | | | | 0.08 | 0.11 | 0.11 | 0.12 |
| | Constant | coef | 3.93 | 4.12 | 4.22 | 3.43 | 2.48 | 2.33 | 2.68 | 5.31 | 2.82 |
| | Constant | std err | 0.10*** | 0.10*** | 0.10*** | 0.12*** | 0.38*** | 0.09*** | 0.33*** | 0.15*** | 0.15*** |
| Region | al Fixed Effects/Dummies | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Nu | Number of Observations | | 864 | 862 | 858 | 864 | 830 | 1714 | 1720 | 1727 | 1686 |
| Adju | Adjusted R-squared (OLS) | | 0.09 | 0.18 | 0.19 | 0.07 | 0.1 | 0.2 | 0.32 | 0.2 | 0.16 |

Table A6: Regressions of Violence Outcomes (on Direct Treatment, within Treated Areas)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level. * significant at 10%; ** significant at 5%; *** significant at 1%.

| | | | | Empowerment 2 | Against Political Vio | lence - Local and i | from the Botton | 1 |
|------------------|-------------------------|--------------|--------------------------------------|---------------------------------|---------------------------------|---|-----------------|-----------------------|
| | Dependent Variable | > | Support for 'Do or Die Affair' | Standing against Violence | Empowerment against Violence | Knowledge of Ways to Counteract Violence | Dissem | (Vote for ination) |
| | | | | | OLS | - | Pro | obit ME |
| | time | coef | -0.67 | 0.56 | 0.63 | 0.59 | | |
| | | std err | 0.18*** | 0.17*** | 0.16*** | 0.20** | | |
| | treat | coef | 0.34 | -0.2 | -0.09 | -0.08 | 0.18 | |
| Main Explanatory | | std err | 0.13** | 0.12 | 0.1 | 0.09 | 0.17 | 0.07 |
| Variables | time*treat | coef | 0 | 0.21 | 0.08 | 0.25 | | |
| | time+treat | std err | 0.1 | 0.11* | 0.12 | 0.12* | | |
| | Constant | coef | 2.52 | 5.34 | 5.27 | 5.22 | -0.58 | |
| | Constant | std err | 0.26*** | 0.15*** | 0.11*** | 0.14*** | 0.16*** | |
| Regiona | l Fixed Effects/Dummies | | Yes | Yes | Yes | Yes | Y | es |
| Nun | Number of Observations | | | 1714 | 1695 | 1711 | 8 | 69 |
| Adjusted R-s | quared (OLS) Pseudo R2 | (P) | 0.24 | 0.24 | 0.27 | 0.25 | 0.04 | |

| 14010110(00144 | ineu): Regressions of | 10101100 | | | , | , | |
|------------------|------------------------|----------|--|---------------------------|---------------------------|--|-------------------------|
| | | | | Crime (| Perceptions and Ex | perience) | 1 |
| | Dependent Variable | > | Thefts Public Place (Perception) | Vandalism (Perception) | Vandalism (Experience) | Psysical Intimidation (Perception) | Problems with Police |
| | | | | | OLS | | |
| | time | coef | -0.48 | -0.27 | -0.06 | -0.32 | -0.41 |
| | | std err | 0.16** | 0.11** | 0.03* | 0.14** | 0.16** |
| | treat | coef | 0 | 0.14 | -0.04 | 0.24 | -0.01 |
| Main Explanatory | treat | std err | 0.08 | 0.08 | 0.04 | 0.11** | 0.17 |
| Variables | time*treat | coef | -0.05 | -0.05 | 0.03 | -0.18 | -0.02 |
| | time-treat | std err | 0.09 | 0.06 | 0.03 | 0.09* | 0.11 |
| ſ | Gundant | coef | 2.9 | 2.18 | 1.06 | 1.86 | 3 |
| | Constant | std err | 0.29*** | 0.15*** | 0.03*** | 0.17*** | 0.33*** |
| Re | Regional Fixed Effects | | | Yes | Yes | Yes | Yes |
| Nui | Number of Observations | | | 1697 | 1727 | 1729 | 1704 |
| Adju | sted R-squared (OLS) | 0.21 | 0.24 | 0.01 | 0.32 | 0.22 | |

Table A6 (Continued): Regressions of Violence Outcomes (on Direct Treatment, within Treated Areas)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

| | | | | Turnout | | | AC in Presidentia | 1 | Incu | mbent in Guberna | torial |
|------------------|------------------------|---------|----------------------|----------------------|--|----------------------|----------------------|--|----------------------|----------------------|--|
| | Dependent Variab | le> | Networks - 'Chat' | Networks - 'Chat' | Networks - 'Chat'/'Family'/' Distance' | Networks - 'Chat' | Networks - 'Chat' | Networks - 'Chat'/'Family'/' Distance' | Networks - 'Chat' | Networks - 'Chat' | Networks - 'Chat'/'Family'/' Distance' |
| | | | | | | | OLS | | | | |
| | time | coef | 0.05 | 0.04 | 0.05 | 0.01 | 0.02 | 0.04 | 0.03 | 0.03 | 0.04 |
| | ume | std err | 0.02*** | 0.01*** | 0.02*** | 0.01 | 0.01* | 0.02** | 0.02* | 0.01* | 0.03 |
| | treat | coef | -0.02 | -0.02 | -0.04 | 0.02 | 0.06 | 0.04 | -0.03 | -0.04 | -0.06 |
| Main Explanatory | treat | std err | 0.02 | 0.02 | 0.02* | 0.02 | 0.02*** | 0.02** | 0.03 | 0.02 | 0.03* |
| Variables | time*treat | coef | 0.05 | 0.06 | 0.04 | -0.01 | -0.02 | -0.04 | 0.06 | 0.07 | 0.08 |
| | time treat | std err | 0.02** | 0.02*** | 0.02* | 0.01 | 0.01* | 0.02* | 0.03* | 0.03** | 0.03** |
| | Constant | coef | 0.75 | 0.73 | 0.69 | 0.2 | 0.17 | 0.17 | 0.48 | 0.44 | 0.35 |
| | Constant | std err | 0.01*** | 0.02*** | 0.03*** | 0.04*** | 0.03*** | 0.03*** | 0.04*** | 0.05*** | 0.05*** |
| Region | nal Fixed Effects | | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Demograp | hic/Political Controls | | No | Yes | Yes | No | Yes | Yes | No | Yes | Yes |
| Numbe | r of Observations | | 1059 | 778 | 1581 | 1080 | 803 | 1541 | 1080 | 985 | 1596 |
| Adju | sted R-squared | | 0.24 | 0.31 | 0.25 | 0.21 | 0.3 | 0.42 | 0.2 | 0.27 | 0.27 |

| | | | | | Но | usehold Size | (*) | | | | | Occupation | : Working a | t Home(*) | | |
|------------------|--|---------|-----------------------|-----------|-----------|--------------|------------------------|---------|--------------------------------|-----------------------|----------|------------|-------------|-----------------------|---------|--------------------------------|
| | Dependent Variable | > | Change in Violence | Change ir | n Turnout | | Vote (AC - lential) | (Incun | in Vote nbent - atorial) | Change in Violence | Change i | n Turnout | 0 | Vote (AC - ential) | (Incun | in Vote nbent - atorial) |
| | | | OLS | Pro | obit | Ordere | d Probit | Ordered | d Probit | OLS | Pr | obit | Ordere | l Probit | Orderee | d Probit |
| | | | | | ME | | ME | | ME | | | ME | | ME | | ME |
| | treatment | coef | 0.88 | 0.06 | 0.02 | -0.71 | -0.13 | 0.41 | 0.16 | 0.6 | 0.19 | 0.06 | -0.35 | -0.07 | 0.3 | 0.12 |
| | treatment | std err | 0.23*** | 0.18 | 0.02 | 0.15*** | -0.15 | 0.18** | 0.10 | 0.16*** | 0.13 | 0.00 | 0.08*** | -0.07 | 0.09*** | 0.12 |
| | treatment*demo(*) | coef | -0.05 | 0.01 | 0 | 0.06 | 0.01 | -0.02 | -0.01 | -0.51 | -0.37 | -0.1 | 0.04 | 0.01 | -0.28 | -0.11 |
| Main Explanatory | treatment (uento()) | std err | 0.02*** | 0.02 | 0 | | 0.01 | 0.02 | -0.01 | 0.25* | 0.25 | -0.1 | 0.26 | 0.01 | 0.29 | -0.11 |
| Variables | demo(*) | coef | 0.02 | -0.01 | 0 | -0.04 | -0.01 | 0.02 | 0.01 | 0.2 | 0.17 | 0.05 | -0.25 | -0.05 | -0.02 | -0.01 |
| | demo(*) | std err | 0.02 | 0.01 | U | 0.01*** | -0.01 | 0.02 | 0.01 | 0.17 | 0.17 | 0.05 | 0.21 | -0.05 | 0.11 | -0.01 |
| | Constant | coef | 0.04 | -0.58 | | | | | | 0.1 | -0.68 | | | | | |
| | Constant | std err | 0.33 | 0.39 | | | | | | 0.33 | 0.38* | | | | | |
| Regional Dum | mies & Demographic/Political C | ontrols | Yes | Y | es | Y | es | Y | es | Yes | Y | es | Y | es | Y | es |
| | Number of Observations | | 965 | 95 | 56 | 11 | .07 | 11 | 00 | 965 | 9 | 56 | 11 | 07 | 11 | 00 |
| Adjusted R-sq | Adjusted R-squared (OLS) Pseudo R2 (P) Chi2 (OP) | | 0.08 | 0.05 | | 1415.85 | | 297.43 | | 0.08 | 0. | 05 | 794 | .67 | 572 | 2.44 |

Table A8: Treatment Interacted with Demographics: are respondents that are affected by the campaign the same across the design?

| | | | | | Pro | perty: Land | (*) | | | | | Ethnic | Group: Yorı | ıba(*) | | |
|------------------|--------------------------------|---------|-----------------------|-----------|-----------|-------------|-----------------------|---------|--|---------|-----------|-----------|---------------------|-----------------------|---------|--------------------------------|
| | Dependent Variable | > | Change in Violence | Change ir | n Turnout | 0 | Vote (AC - ential) | (Incun | Change in Vote (Incumbent - Gibernatorial) | | Change in | n Turnout | Change in Presid | Vote (AC - ential) | (Incun | in Vote 1bent - atorial) |
| | | | OLS | Pro | bit | Ordere | d Probit | Ordered | l Probit | OLS | Pro | obit | Ordere | d Probit | Ordered | l Probit |
| | | | | | ME | | ME | | ME | | | ME | | ME | | ME |
| | treatment | coef | 0.79 | 0.1 | 0.03 | -0.5 | -0.09 | 0.28 | 0.11 | 0.22 | 0.18 | 0.05 | -0.24 | -0.04 | 0.27 | 0.1 |
| | treatment | std err | 0.21*** | 0.13 | 0.05 | 0.14*** | -0.09 | 0.12** | 0.11 | 0.12* | 0.14 | 0.05 | 0.08*** | -0.04 | 0.12** | 0.1 |
| | treatment*demo(*) | coef | -0.46 | 0.09 | 0.03 | 0.29 | 0.05 | 0 | 0 | 1.12 | -0.13 | -0.04 | -0.32 | -0.06 | 0.03 | 0.01 |
| Main Explanatory | | std err | 0.18** | 0.18 | 0.03 | 0.14** | 0.05 | 0.17 | U | 0.27*** | 0.24 | -0.04 | 0.15** | -0.00 | 0.23 | 0.01 |
| Variables | demo(*) | coef | 0.26 | -0.28 | -0.08 | -0.23 | -0.04 | -0.03 | -0.01 | -0.52 | 0.23 | -0.07 | -0.24 | -0.04 | -0.1 | -0.04 |
| | denio(*) | std err | 0.15* | 0.14** | -0.08 | 0.08*** | -0.04 | 0.14 | -0.01 | 0.35 | 0.22 | -0.07 | 0.22 | -0.04 | 0.15 | -0.04 |
| | Constant | coef | 0.01 | -0.6 | | | | | | 0.28 | -0.64 | | | | | |
| | Constant | std err | 0.33 | 0.37 | | | | | | 0.33 | 0.37* | | | | | |
| Regional Dum | mies & Demographic/Political C | ontrols | Yes | Y | es | Y | es | Y | es | Yes | Y | es | Y | es | Y | es |
| | Number of Observations | | 965 | 95 | 56 | 11 | 07 | 10 | 99 | 965 | 9. | 56 | 11 | 07 | 11 | 00 |
| Adjusted R-sq | uared (OLS) Pseudo R2 (P) Chi | 2 (OP) | 0.08 | 0. |)5 | 680 |)2.7 | 295 | .31 | 0.11 | 0. | 05 | 102 | 8.22 | 62: | 5.3 |

Note: Standard errors reported; these are corrected by clustering at the location (census area) level.

| | Dependent Variable> | | Conflict Community | Gang Activity | Security | Intensity (Journals) | Empower | Knowledge | ge Postcard | | Physical Intimidation |
|------------|--|---------|-----------------------|------------------|----------|-------------------------|----------|-----------|-------------|---------|--------------------------|
| | | | OLS . | | | | | | Probit | | OLS |
| | | | | 015 | | | | | | ME | ULS |
| | time | coef | -0.52 | -0.18 | 0.14 | -0.42 | 0.53 | 0.68 | | | -0.04 |
| | | std err | 0.17** | 0.21 | 0.16 | 0.16** | 0.15*** | 0.14*** | | | 0.15 |
| | distance time*distance Constant | coef | -0.0042 | -0.0078 | 0.0013 | -0.0044 | -0.0018 | -0.0013 | -0.0063 | -0.0022 | -0.0095 |
| | | std err | 0.0019** | 0.0030** | 0.0024 | 0.0027 | 0.0021 | 0.0015 | 0.0056 | | 0.0046* |
| | | coef | 0.0032 | 0.0025 | -0.0010 | 0.0108 | -0.0051 | -0.0053 | | | -0.0003 |
| | | std err | 0.0037 | 0.0038 | 0.0032 | 0.0029*** | 0.0018** | 0.0020** | | | 0.0030 |
| | | coef | 1.63 | 4.11 | 3.86 | 2.63 | 3.94 | 4.98 | -0.03 | | 3.12 |
| | Constant | std err | 0.29*** | 0.84*** | 0.23*** | 0.1107*** | 0.22*** | 0.20*** | 0.22 | | 0.92*** |
| F | Regional Fixed Effects | | | Yes | Yes | Yes | Yes | Yes | Yes | | Yes |
| E | Demographic Controls | | | Yes | Yes | Yes | Yes | Yes | Yes | | Yes |
| N | Number of Observations | | | 1152 | 1152 | 64 | 1134 | 1146 | 5 | 78 | 1154 |
| Adjusted F | Adjusted R-squared (OLS) Pseudo R2 (P) | | | 0.27 | 0.1 | -0.07 | 0.23 | 0.21 | 0. | 14 | 0.09 |

 Table A9: Regressions of Outcomes on Distance to Closest Treatment Area (Control Locations Only)

Note: Standard errors reported; these are corrected by clustering at the location (census area) level. * significant at 10%; ** significant at 5%; *** significant at 1%.

| | Dependent Variable> | | | Turnout | | lential C | Gubernatorial Incumbent | | |
|------------|--|---------|---------|---------|---------|--------------|----------------------------|---------|--|
| | | | Probit | | | | | | |
| | | | | ME | | ME | | ME | |
| | time | coef | 0.34 | 0.10 | 0.41 | 0.06 | 0.25 | 0.10 | |
| | ume | std err | 0.13*** | 0.10 | 0.20** | | 0.17 | | |
| | distance | coef | -0.0008 | -0.0003 | 0.0023 | 0.0003 | 0.0109 | 0.0043 | |
| | uistance | std err | 0.0018 | -0.0003 | 0.0068 | 0.0003 | 0.0040*** | | |
| | time*distance | coef | 0.0045 | 0.0014 | -0.0043 | -0.0007 | -0.0036 | -0.0014 | |
| | time ustance | std err | 0.0040 | | 0.0036 | | 0.0032 | | |
| | Constant | coef | 0.12 | | -0.64 | | -0.59 | | |
| | Constant | std err | 0.19 | | 0.25** | | 0.2283*** | | |
| R | Regional Fixed Effects | | | Yes | | Yes | | Yes | |
| D | Demographic Controls | | | Yes | | Yes | | Yes | |
| Nu | Number of Observations | | | 1161 | | 983 | | 1176 | |
| Adjusted R | Adjusted R-squared (OLS) Pseudo R2 (P) | | | 0.06 | | 0.18 | | 0.09 | |